

FIRST NATIONS UNIVERSITY OF CANADA

October 1, 2014 - September 30, 2018

COLLECTIVE AGREEMENT

CANADIAN UNION OF PUBLIC EMPLOYEES Local 5791



TABLE OF CONTENTS

DEFINITIONS		11
ARTICLE 1 PREAMBLE		12
1.1	Working Agreement	12
1.2	Terms and Conditions	12
1.3	Previous Provisions	12
1.4	Conflicting Laws	12
1.5	Employer Rights	13
ARTICLE 2	2 RECOGNITION	13
2.1	Bargaining Agent	13
2.2	Work of the Bargaining Unit	13
2.3	Contracting Out	13
2.4	No Other Agreements	14
2.5	Representatives of CUPE	14
2.6	Union-Management Committee	
2.6.1	Composition of the Committee	14
2.6.2	Purpose	14
2.6.3	Meeting Requests	
2.6.4	Non-Action Grievable	
2.6.5	Implementation of Awards	15
2.6.6	Exchange of Information	15
	UNION ACTIVITIES AND USE OF UNIVERSITY PREMISES	
3.1	Time Off for Union Meetings	
3.2	Time Off for Union Activities	
3.2.1	0.0.0	
3.2.2	Joint Committees	
3.2.3	Executive Member-at-Large	
3.2.4	Employees	16
3.3	Use of University Facilities	16
3.4	Space for Meetings	16
3.5	Bulletin Boards	16
ARTICLE 4	DUES, MEMBERSHIP AND EMPLOYEE INFORMATION	16
4.1	Union Dues	16
4.1.1		
4.1.2	Dues Receipts	17
4.2	CUPE Membership	17
4.2.1	Condition of Employment	17

4.2.2	Provision for Exclusion	17
4.3	New Employees	17
4.4	Visit to the Union Office – New Employees	17
4.5	Employee Information	18
4.5.1		
4.5.2		
4.5.3	Contact Information	18
4.6	Access to Employee Files	19
ARTICLE	5 Harassment/EMPLOYMENT EQUITY	19
5.1	Harassment Commitment	19
5.2	Employment Equity	19
5.3	Duty to Accommodate	20
5.4	Open Communication	20
ARTICLE	6 MECHANISMS FOR DISPUTE RESOLUTION	20
6.1	Handling Differences	20
6.2	Stage 1 - Informal Process	20
6.3	Stage 2 - Formal Grievance Process	21
6.4	Arbitration	
6.4.1		21
6.4.2		
6.4.3		22
6.4.4 6.4.5	· · · · · · · · · · · · · · · · · · ·	
6.4.6		
6.4.7		22 23
6.5		23
6.5.1	Recognition of Stewards and Designated Union Representatives	23
6.5.2		
6.5.3		
6.5.4		23
6.5.5	Authorized Acting Officials	23
6.5.6		
6.5.7	Copies of Documents	24
ARTICLE	7 TYPES OF POSITIONS AND EMPLOYEE PROVISIONS	24
7.1	Full-Time	24
7.2	Regular Part-Time	24
7.3	Permanent	24
7.4	Term	24
7.5	Student Worker Positions	
7.5.1	Student Worker Positions	26

7.5. 7.5.		26 26
7.6	Casual Employees	26
ARTICLE	8 SENIORITY	27
8.1	Seniority Defined	27
8.2	Special Adjustments to Seniority	27
8.2.3	1 Temporary Employment Within Another University Bargaining Unit	27
8.2.2		
8.2.3	3 Resignation	27
8.3	Seniority Rosters	28
8.4	Correction of Seniority Roster	28
ARTICLE	9 VACANCIES, APPOINTMENTS AND PROBATION	28
9.1	Posting	28
9.2	Bidding on Vacant Positions	28
9.3	Notice of Results	29
9.4	Basis for Selection	29
9.5	Provisional Period	29
9.5.3		
9.5.2	2 Reversion	30
9.6	On-the-Job Training	31
9.7	Probationary Period	31
9.7.3	Extension Due to Absence	31
9.7.2		31
9.7.3	3 Termination During Probationary Period	31
9.8	Length of Probationary Periods	32
ARTICLE	10 WORKLOAD ISSUES	
10.1	Requests for Review	32
10.2	Dispute Resolution	32
ARTICLE	11 – HOURS OF WORK AND SPECIAL PAY PROVISIONS	32
11.1	Daily Hours of Work	32
11.2	Part-Time Employees	
11.3	Accommodation of Employees Who Are Students	
11.4	Overtime	34
11.4		34
11.4		34
11.4		
11.4 11.4		34 35
11.4 11.4		35 35
11.4		

11.4	4.7 Call Out	
	1.7.1 Overtime	35
	1.7.2 Scheduled Employees	
11.5	Unsocial Hours of Work Premium	
11.6	Provision of Meal	36
11.7	Childcare Expenses	
11.8	Provision of Tools	
11.9	Bilingual Bonus	
11.10		
11.11	Receipted Expenses	37
11.12	Employment Travel Expense	37
11.13	Itemized Statement	37
11.14	· · · · · · · · · · · · · · · · · · ·	
	14.1 Within the Bargaining Unit 14.2 Outside the Bargaining Unit	
11.15		
ARTICLE	12 PERFORMANCE EVALUATION AND INCREMENTS	
12.1	Process and Responsibilities	
12.2	Performance Evaluation Grievances	
12.3	Changes to Anniversary Date	
12.4	Working Service	39
12.5	Financial Exigency	40
12.6	Preferential Treatment	40
ARTICLE	13 DISCIPLINE	40
13.1	Principle of Innocence and Burden of Proof	
13.2	Written Reprimand	40
13.3	Written Warning	41
13.4	Suspension	41
13.5	Dismissal	41
13.6	Benefits During Suspension	42
ARTICLE	14 RESIGNATION	42
ARTICLE	15 RETIREMENT	42
15.1	Retirement Age and Benefit Coverage	
15.2	Special Arrangements	42

ARTICLE 1	6 – ENTRY AND RE-ENTRY FROM EXCLUDED POSITIONS	43
ARTICLE 1	7 – STAFF REDUCTION	44
17.1	Determination	44
17.2	Notice	44
17.2.1		44
17.2.2	Notice To the Employee	44
17.2.3		
17.3	Assisted Early Retirement	46
17.3.1		46
17.3.2		
17.4	Redeployment and/or Retraining	47
17.4.1		47
17.4.2	Provisional Period	47
17.5	Termination and Severance	48
17.5.1		
17.5.2	Termination Date	48
17.5.3	Calculation of Severance Pay	48
17.6	Bumping	48
17.6.1	Notification by the Employee	48
17.6.2		48
17.7	Layoff and Recall	49
17.7.1	Applications for Re-Employment	49
17.7.2	Election to Work Casual or Term	50
17.7.3	Reporting for Duty	50
17.7.4		50
17.7.5		50
17.7.6		50
17.7.7		51
17.7.8		
17.7.9	Grievances	
ARTICLE 1	8 CLEARANCE UPON TERMINATION	51
ARTICLE 1	9 PAYMENT TO ESTATES OF DECEASED EMPLOYEES	51
ARTICLE 2	0 HOLIDAYS	52
	Named Holidays	
	Holiday Pay	
	Observation of Named Holidays	
	Compensation for Work on a Holiday	
	Closure Due to Inclement Weather or Other Emergency	
ARTICLE 2	1 – ANNUAL LEAVE	
21.1	Accumulation	53

21.2	Annual Leave Year	53
21.3	Special Circumstances	54
21.4	Time of Annual Leave	54
21.5	Named Holiday During Annual Leave	54
21.6	Annual Leave for Shift Employees	54
21.7	Approved Absence During Annual Leave	
21.8	Requests For Annual Leave	54
ARTICLE	22 EDUCATION LEAVE AND CAREER DEVELOPMENT	
22.1	Recognition	
22.2	Career Counselling	
22.3	Taking Classes	
22.3	8.1 University Classes	55
22.3	Classes Other than University Classes	56
22.4	Education Leave	56
22.5	Professional Development Leave	57
22.5		57
22.5	P - /	
22.6	Financial Assistance	
ARTICLE	23 EMPLOYEE PENSION AND BENEFITS PLANS	58
23.1	Pension Plan	59
23.1	.1 Payment of Contributions	59
23.1		
23.2 23.2	Group Insurance	59
23.2		
23.3	Dental and Extended Health and Vision Care Benefits Plans	
23.4	Employee Benefits During Leave of Absence Without Pay	
23.4	1.1 Leave Without Pay for Four Consecutive Weeks or Fewer	60
23.4	2 Leave Without Pay for More Than Four Consecutive Weeks	60
23.5	Coverage	60
23.6	Annual Statement	61
23.7	Employee Family Assistance Program (EFAP)	61
ARTICLE	24 – RE-ENTRY INTO THE WORKPLACE (AFTER LONG-TERM DISABILITY)	61
ARTICLE	25 INSURANCE	61
25.1	Liability Insurance	61
25.2	Travel Insurance	61

ARTICLE	26 LEAVE FOR COURT, INCARCERATION OR JURY DUTY	62
26.1	Leave for Court Appearance or Incarceration	62
26.2	Leave for Jury Duty or Court Summons	62
Article 2	7 POLITICAL OFFICE, UNION AND SECONDMENT LEAVE	62
27.1 27.1 27.1 27.1 27.1	Lyon Election to Hold Office Arrow After Term in Office	62 62 63
27.2	Secondment Leave	63
27.3	Casual Union Leave	63
ARTICLE MONTH	28 – MATERNITY/ADOPTION/PARENTAL LEAVE OR CHILD IN CARE UNDER 18 S	64
28.1	Service Requirements for Maternity/Adoption/Parental Leave/Child in Care	
28.2	Length of Maternity/Adoption/Parental Leave/Child in Care	64
28.3	Supplementary Employment Benefits	64
28.4	Seniority Status During Maternity/Adoption/Parental Leave/Child in Care	64
28.5	Return from Maternity/Adoption/Parental Leave/Child in Care	64
ARTICLE	29 SICK LEAVE	65
29.1	Sick Leave Defined	65
29.2	Rate of Accumulation	65
29.3	Application for Disability Benefits	65
29.4	Entitlement on Termination	65
29.5	Illness in the Family	65
29.6	Notification and Proof of Illness	65
29.7	Time Off for Medical Examinations	65
29.8	Absence in Excess of Sick Leave Credits	66
29.9	Utilization of Unearned Sick Leave Credits	66
29.10	Utilization of Earned Sick Leave Credits While on Short Term Disability	66
29.11	Pressing Necessity Leave	66
ARTICLE	30 COMPASSIONATE LEAVE	66
30.1	Grave Illness or Death - Immediate Family	66
30.2	Death of Other Family Members	66
30.3	Additional Days	66

30.4	Extended Leave of Absence – Care of a Family Member	67
ARTICLE	31 OTHER LEAVES	67
31.1	Special Leave of Absence	67
31.2	Leave for Band Elections	67
31.3	Leave for Shift Workers	68
31.4	Leave for Traditional Ceremonies	68
31.5	Personal Leave	68
31.6	Religious Holidays	68
ARTICLE	32 OCCUPATIONAL HEALTH AND SAFETY	68
32.1	Co-operation on Safety	68
32.2	Role of the Committee and Pay Provisions	68
32.3	Safety Measures	69
32.4	Educational and Training Programs	69
32.5	Safety and Health Reports, Records and Data	70
32.6	Right to Refuse	70
32.7	Security	70
ARTICLE	33 COPYRIGHTS AND PATENTS	70
ARTICLE	34 NO STRIKES OR LOCKOUTS	71
34.1	No Strikes or Lockouts	71
34.2	Crossing Picket Lines During a Strike and Handling Goods	71
ARTICLE	35 DURATION AND CONTINUATION OF AGREEMENT	71
35.1	Duration	71
35.2	Expiration	71
35.3	Negotiations	71
ARTICLE	36 GENDER NEUTRAL JOB EVALUATION PLAN MAINTENANCE	71
36.1	Purpose	71
36.2	Pay Grades	72
36.3	Composition of Committee	72
36.4	Time Off	72
36.5	Resolution of Differences	72
36.6	New and Existing Positions	
36. 36.		72 72
36.		72

36.7 Process	72
36.7.1	72
36.7.2	73
36.8 Requests for Reclassification	73
36.8.1 Review Process for Requests	73
36.8.2 Conduction of the Review 36.8.3 Salary Adjustment Upon Reclassification	
ARTICLE 37 WAGE	74
ARTICLE 38 Job-Sharing	74
38.1	74
38.2	74
38.3	74
38.4	
38.5	
38.6	
ARTICLE 39 JURISDICTION OF NEW AND REVISED POSITIONS	75
ARTICLE 40 FUNCTIONAL SUPERVISION	76
SCHEDULE A – FIRST NATIONS UNIVERSITY OF CANADA CUPE 1975 SALARY GRID	77
SCHEDULE B	79
APPENDIX A: HARASSMENT & DISCRIMINATION PREVENTION RESOURCES	88
APPENDIX B: DEFINITIONS AND EXAMPLES OF ATTITUDINAL HARASSMENT	89
APPENDIX C: DEFINITION AND EXAMPLES OF SEXUAL HARASSMENT	90
APPENDIX D: DEFINITION AND EXAMPLES OF RACIAL HARASSMENT	92
SCHEDULE C	93
SCHEDULE D	94
SCHEDULE E – LETTERS OF UNDERSTANDING	97
MEMORANDUM OF AGREEMENT	106
SIGNING PAGE	108

DEFINITIONS

For the purposes of this Agreement:

"Bargaining Unit" shall refer to the First Nations University of Canada, Canadian Union of Public Employees group of employees as defined by the Certification Order of the Saskatchewan Labour Relations Board issued in Regina on October 8, 1996 or as may be amended from time to time by the said board or by mutual agreement of the parties to this Agreement.

"Employee(s)" shall refer to any persons defined and covered by this Agreement. Employees are entitled to all rights and benefits of this Agreement unless otherwise specified in this Agreement.

"Union representative(s)" shall refer to any employee duly authorized by the Union. The term shall refer to employees who are elected, appointed or who volunteer to fill Union positions that represent the First Nations University of Canada, Canadian Union of Public Employees group of employees on committees, as Stewards or on the CUPE 5791 - 01 Executive Board.

"Employer" shall refer to the First Nations University of Canada.

ARTICLE 1 -- PREAMBLE

1.1 Working Agreement

The parties to this Collective Agreement (herein after referred to as the "Agreement") recognize that the First Nations University of Canada (formerly known as the Saskatchewan Indian Federated College) is a unique, First Nationscontrolled educational institution whose objectives include service to First Nations communities and the promotion, preservation, protection and interpretation of First Nations histories, languages, cultures and artistic heritages using First Nations and non-First Nations ways of knowing and understanding. As well, the First Nations University of Canada is founded upon and operates with the guidance and blessing of First Nations Elders.

The First Nations University of Canada embodies the values and aspirations of First Nations people to achieve education of quality within a uniquely First Nations environment and seeks to promote, for all members of the University family, the development of spirit, body, mind and sense of community.

The parties to this Agreement agree to work together toward these goals to promote harmonious relations and to settle misunderstandings and disagreements peacefully in the spirit of the traditional First Nations values of tolerance, humility, mutual respect and sharing.

1.2 Terms and Conditions

This Agreement sets forth the terms and conditions governing First Nations University of Canada (hereinafter referred to as the "employer") employees represented by the Canadian Union of Public Employees (CUPE) (hereinafter referred to as the "Union") and is binding upon the parties to this Agreement.

Both parties agree to:

- a) affirm the principle of First Nations jurisdiction over the University; and
- b) support and encourage the rights of First Nations to exercise their inherent right to govern their own affairs.

1.3 Previous Provisions

Nothing in this Agreement shall affect any provisions or concessions in existence at the time of signing this Agreement, which are more favourable to employees than those contained in this Agreement.

1.4 Conflicting Laws

If any provision of this Agreement or of any agreement made in pursuance thereof is found to be contrary to the provisions of any law, now or hereafter enacted, this Agreement will not be abrogated but will be subject to such amendments as may be necessary to bring it into conformity with the law.

1.5 Employer Rights

All the functions, rights, powers and authority, which the employer has not specifically abridged, delegated or modified by this Agreement, are recognized by the Union as being retained by the employer.

The employer shall exercise this right in a manner that is fair, non-arbitrary and reasonable.

ARTICLE 2 -- RECOGNITION

2.1 Bargaining Agent

The employer recognizes the Union as the exclusive bargaining agent of the members of the bargaining unit as defined by the Certification Order of the Saskatchewan Labour Relations Board issued in Regina on October 8, 1996 or as may be amended from time to time by the said Board or by mutual agreement of the parties to this Agreement.

The employer agrees to negotiate with the Union and any of its duly authorized Union representatives or committees, concerning all matters affecting the relationship between the parties to this Agreement, aiming toward a peaceful and amicable settlement of any differences that may arise between the parties to this Agreement.

The employer agrees, during the term of this agreement, to bring all positions falling within the scope of this agreement into this agreement, including all CUPE equivalent research funded positions and any personnel not currently covered by URFA and APT.

The parties agree to jointly submit a new certification order.

2.2 Work of the Bargaining Unit

Employees of the University whose jobs are not in the bargaining unit shall not regularly work on any jobs, which are included in the bargaining unit unless mutually agreed upon by the parties to this Agreement. Agreements of this nature will include a list of duties and a time frame.

2.3 Contracting Out

(a) In order to provide job security for the members of the bargaining unit, the employer agrees that it will not reduce pay or benefits or lay off employees in order to contract out the duties normally performed by members of the bargaining unit; nor will the employer replace vacant positions by contracting out work which is normally performed by the bargaining unit. (b) The employer agrees that members of the CUPE 5791 bargaining unit will be the cleaners/janitors and maintenance workers for all future tenants of the First Nations University of Canada.

2.4 No Other Agreements

No employee(s) shall be required or permitted to make a written or verbal agreement with their supervisor, employer or representatives of the employer which conflict with the terms of this Agreement.

In order that this may be carried out, the Union will supply the employer with the names of its Union representatives and the employer will supply, at any time it is requested in regard to any employee or group of employees, the names of their supervisors and functional supervisors. (See Article 41)

2.5 Representatives of CUPE

The Union shall have the right at any time to have the assistance of representatives of the Canadian Union of Public Employees when dealing or negotiating with the employer. Such representative(s) shall also have access to the employer's premises in order to investigate and assist in the settlement of grievances.

2.6 Union-Management Committee

2.6.1 Composition of the Committee

There shall be a Union-Management Committee consisting of three (3) Union representatives and two (2) employer representatives for the purpose of resolving difficulties and promoting harmonious relationships.

2.6.2 Purpose

The purpose of the Union-Management Committee shall be to discuss and settle, if possible, matters of mutual concern. In matters that are personal or particularly sensitive, strict confidentiality shall be maintained.

2.6.3 Meeting Requests

At the request of either party, the Committee shall meet within fifteen (15) calendar days at a time and place fixed by mutual agreement. The request must include the reason(s) for the meeting.

2.6.4 Non-Action Grievable

Failure on the part of the employer to complete an action agreed upon in meeting and in a timely manner, shall be grievable.

2.6.5 Implementation of Awards

In addition to the foregoing, the implementation of arbitration awards and final court decisions relating thereto will be discussed at the request of either party.

2.6.6 Exchange of Information

The employer and the Union agree to exchange at the request of either party, information that is not confidential, is readily available and pertains to a matter under discussion between the parties.

All correspondence between the employer and an employee shall be copied to the Union.

ARTICLE 3 -- UNION ACTIVITIES AND USE OF UNIVERSITY PREMISES

3.1 Time Off for Union Meetings

The employer shall allow each employee an equivalent of seven (7) hours off per year with pay for Union meetings.

3.2 Time Off for Union Activities

3.2.1 Bargaining Committee

The employer shall recognize an employee Bargaining Committee of up to three (3) employees who hold positions in different departments.

The Union will advise an employee's supervisor that the employee will be serving on the Bargaining Committee. Upon notifying their supervisor, Bargaining Committee employees shall have reasonable time off with pay for the purpose of preparing for and attending bargaining sessions.

3.2.2 Joint Committees

The employer will provide reasonable time off with pay to employees as necessary to fulfill their responsibilities as Union representatives on joint committees addressed in this Agreement and any other committees that may, from time to time, be jointly organized by the employer and the bargaining unit.

3.2.3 Executive Member-at-Large

The employer will provide reasonable time off without pay to the bargaining unit's Executive Member-at-Large to attend Union business and educational courses, workshops and conventions as required with

the understanding that the Executive Member-at-Large also serves as Steward. (See Article 6.8.2)

3.2.4 Employees

The employer will provide reasonable time off without pay to employees to attend Union educational courses, workshops and conventions.

3.3 Use of University Facilities

The employer shall provide the bargaining unit with reasonable private and secure office space on the premises, reasonable free use of office supplies, duplicating services, computing facilities, audio-visual and video conferencing equipment, IT support and the University internal, Canada Post, courier and electronic mail services.

3.4 Space for Meetings

The employer shall allow the Union to hold meetings and educational functions, and to conduct Union business at the University. The employer will make space available for such meetings and functions subject to normal scheduling restrictions.

3.5 Bulletin Boards

The employer shall provide Union bulletin boards, which shall be placed so that all employees will have access to them and upon which the Union shall have the right to post notices of meetings and such other notices as may be of interest to Union employees.

The employer will provide web space for placement of the FNUniv-CUPE 5791 Collective Bargaining Agreement on the Human Resources Department web site.

ARTICLE 4 -- DUES, MEMBERSHIP AND EMPLOYEE INFORMATION

4.1 Union Dues

4.1.1 Deduction of Union Dues

Upon obtaining written authorization from the employee, the employer shall deduct, as a condition of employment, initiation fees, dues and such other assessments (subject to Article 4.2) as the Union may direct in writing through its Secretary-Treasurer, from each employee's monthly salary and remit the same prior to the 15th day of the following month to the Secretary-Treasurer of CUPE Local 5791-01 accompanied by a list of names of all employees, the amount of dues deducted on behalf of each employee for the month and the total of regular wages paid in the month.

4.1.2 Dues Receipts

At the time that Income Tax (T4) slips are made available, the employer shall include information on the amount of Union dues paid by each CUPE member in the previous year that is deductible for income tax purposes, subject to receipt of certification satisfactory to Revenue Canada.

4.2 CUPE Membership

4.2.1 Condition of Employment

Every employee who is now or hereafter becomes a member of the Union shall maintain membership in the Union as a condition of employment. Every new employee whose employment commences hereafter shall, within 30 days after commencing employment, apply for and maintain membership in the Union and maintain membership in the Union as a condition of employment, provided that any employee in the bargaining unit who is not required to maintain membership or apply for and maintain membership in the Union shall, as a condition of employment, tender to the Union the periodic dues uniformly required to be paid by the members of the Union.

4.2.2 Provision for Exclusion

An employee may be excluded from the provisions of Article 4.2.1 if the employee objects to joining or belonging to a trade union as a matter of conscience, religious training or religious belief. In such case, the employee shall pay an amount equal to Union dues to a charity mutually agreed by the employee and CUPE Local 5791-01. If there is a dispute concerning an employee objection, such dispute shall be referred to the Labour Relations Board in accordance with Section 5 (1) of *The Saskatchewan Employment Act*.

4.3 New Employees

The employer shall inform each new member of the bargaining unit of the provisions of Article 4.1 not later than 30 days after the employee's date of appointment, and at the same time, provide the member of the bargaining unit with a paper copy of the Agreement and the link to the collective agreement on the CUPE 5791 website.

4.4 Visit to the Union Office - New Employees

Each new employee will be given the opportunity during the first month of employment, within regular working hours, of visiting the Union Office or meeting with a Steward for the purpose of joining the Union and becoming acquainted with the rights and responsibilities of membership. Such absence from the work place will be reasonably brief and taken at a time convenient to both the employee and the supervisor.

4.5 Employee Information

4.5.1 Reporting to the Union

The employer shall provide the Union with a copy of the Letter of Offer (or revision thereof) of any newly hired employee or the revised Letter of Offer of any current employee within fifteen (15) days of receipt of the signed Letter of Offer. This Letter of Offer will include the employee's status, position, pay grade and salary.

4.5.2 Employee Lists

On the twentieth (20th) day of each month, the employer shall make available to the Union a list of all employees in the **CUPE 5791** bargaining unit.

The list shall include the following:

- name;
- gender;
- employment status (i.e.: permanent, term, full-time, part-time);
- position title;
- pay grade;
- department;
- date of first appointment to a bargaining unit position;
- date appointed to current position;
- start and end dates of term appointments;
- retirement date;
- start date with the University;
- leaves and types of leaves;
- any vacant bargaining unit position(s);
- for casual employees: mailing address, position(s) filled, dates and total hours worked during the month for each position filled.

4.5.3 Contact Information

The employer will provide to the Union a list of all employees in the bargaining unit. The list will include each person's name, position title, home mailing address, home telephone number and work email.

The list will also include the employee's work status (permanent, term, full-time, part-time, casual, student workers), and if the employee is on leave of absence, the nature of the leave.

The employee contact list will be provided on an electronic spreadsheet to the Union by September 1 of each year.

4.6 Access to Employee Files

There shall be only one official file for each employee and it shall be located in the Human Resources Department. The file shall contain only material pertinent to the employee's employment. It shall not contain anonymous material. The employee may add a signed and dated response to any document contained therein.

An employee (or other person with written permission from the employee) shall have the right to examine the official file at any time during regular office hours mutually agreeable to the employee and the Human Resources Department in the company of a human resources employee. The employee may be accompanied, if desired, by a Steward or Union representative of the employee's choosing.

An employee (or other person with written permission from the employee) shall be provided with a copy of a reasonable number of documents in the file.

The employee's supervisor may have a working file of material destined for the official file and it shall be subject to the same provisions as the official file.

The employer shall also maintain a payroll file for each employee and it shall be located in the Finance Department. The file shall contain only material pertinent to the employee's financial affairs with the University. An employee shall have the right to examine records in their payroll file on the same terms and conditions as enumerated above for the official file.

ARTICLE 5 -- HARASSMENT/EMPLOYMENT EQUITY

5.1 Harassment Commitment

The employer is committed to maintaining a mutually agreed upon policy that will prohibit harassment and ensures that all members of the University's community have the right to participate equally in activities at the University without fear of harassment and providing an environment which affirms and promotes First Nations cultures and values, and the dignity of human beings of diverse backgrounds and needs.

The parties agree that, the University Policy: Respectful University (or any amended policy thereof) does not discourage, prevent or preclude CUPE members from filing a grievance.

5.2 Employment Equity

The parties agree that there are legitimate reasons why the emphasis on employment and training opportunities for Canadian First Nations people is necessary to ensure a strong First Nations presence within the University. The parties recognize the employment equity exemption granted the University by the Human Rights Commission and agree that hiring consistent with this exemption does not contravene this Agreement. When recruiting new employees, the employer shall ensure that all advertisements reflect the employer's commitment to the hiring of Canadian First Nations and employment equity.

5.3 Duty to Accommodate

The University and the Union recognize their joint duty under The Saskatchewan Human Rights Code to provide accommodation in the workplace to employees. Such accommodation shall be provided up to the point of undue hardship for the University.

5.4 Open Communication

The University and the Union recognize the value of open communication as it promotes positive human relations and a productive work environment. To that end, the University will endeavor to provide updates on decisions and University developments. In addition, all policies and procedures will be made available to Union members as they are approved.

ARTICLE 6 -- MECHANISMS FOR DISPUTE RESOLUTION

The time limits imposed by the following provisions may be waived by written agreement between the employer and the Union.

6.1 Handling Differences

Whenever a difference shall arise between the parties hereto or any person bound by this Agreement concerning its interpretation, application, operation or alleged violation, the difference shall be dealt with in accordance with the processes herein outlined.

Any processes prescribed in this Agreement, which contain a different dispute resolution process binding on both parties than what is contained in Article 6, shall be governed by that process.

6.2 Stage 1 - Informal Process

Before a grievance is filed, an attempt will be made to settle the issue by informal discussion. An employee(s), a Steward or designated Union Representative may, at any time, present a verbal complaint to their out-of-scope supervisor, or equivalent, and/or the Human Resources Department. Employees accessing the informal process shall have union representation at any meetings with the employer to discuss the complaint.

If the matter is not satisfactorily settled, the Union may present the complaint for resolution at the next monthly Union Management Committee

meeting or a Talking Circle. The affected employee(s) may attend the portion of the meeting allocated to the complaint. Depending on the situation, the Union Management Committee may also request the attendance of the employee's out-of-scope supervisor or other involved person(s). An employee may also request the presence of an Elder.

Employees presenting complaint and/or attending follow up meetings and/or discussion with the Union do so with the full cooperation of their supervisor and without loss of pay.

6.3 Stage 2 - Formal Grievance Process

Failing resolution at Stage I - Informal Process, the Union may, within thirty (30) calendar days, of a written decision arising from the Joint Union Management Meeting or failure to arrive at consensus at the Talking Circle, refer the written grievance to the Director, Human Resources.

The Steward shall refer the written grievance signed by both the Steward representing the employee(s) concerned and the employee(s), where applicable, to the Human Resources Department. The written grievance will include details of the incident in dispute, which article(s) of the Agreement has allegedly been violated, and a recommended remedy(s) to correct the alleged violation.

The filing of a grievance is a call for a meeting to discuss the grievance and is to be held within fifteen (15) calendar days at a time and place fixed by mutual agreement.

One or more of the aggrieved parties may be present at the meeting at the request of either party. Employees attending a meeting shall do so without loss of pay.

6.4 Arbitration

6.4.1 Time Limit

In the event that any grievance or matter in dispute has not been settled through the procedure outlined above, either party may, within thirty (30) calendar days, submit the grievance or matter in dispute to an Arbitration Board in accordance with Article 6.5.

The time limits imposed by the following provisions may be waived by agreement between the employer and the Union.

6.4.2 Composition of the Board

The Arbitration Board shall consist of three members. One shall be named by the employer and one named by the Union. Each of the parties to this Agreement shall have their respective Board member selected and made known to each other within seven (7) working days of notice being given by either party for the establishment of the Board. The third position on the Board is designated the Chair.

6.4.3 Selection of the Chair

The parties shall endeavour to agree on a Chair within twenty-one (21) working days of the grievance being submitted for arbitration. In the event of their failure to agree, either party may notify the Chair of the Saskatchewan Indian Personnel Commission who shall select by lot a Chair from a list of names agreed by the parties. Once appointed, the Chair shall invite a First Nations University Elder to act as her/his advisor. The Elder shall guide the processes along traditional First Nations methods of conflict resolution.

6.4.4 Hearing

The Board, having been formed by the above procedure, shall meet, hear the evidence of both parties, and render a written decision within sixty (60) calendar days from the completion of taking evidence. The Board may determine its own procedure, in consideration of First Nations traditions and values, but will ensure that each party has the opportunity to present its case. At the hearing, each party shall be entitled to be represented by counsel or otherwise, to present evidence, to cross examine the witnesses of the other party and to present arguments either orally or in writing.

The decision of the majority of the Board on the matter at issue shall be final and binding on both parties. The Arbitration Board shall not have the power to alter any of the terms of this Agreement or to substitute any provisions, nor to give any decision inconsistent with the terms of this Agreement.

6.4.5 Time Deficiencies

Any Board of Arbitration established pursuant to the grievance procedure shall have the power to hear any arguments as to whether, in order to avoid consideration of substantive issues, time limits set forth in the grievance procedure have been unreasonably enforced. The Board may decide to deal with the case placed before it despite such minor time deficiencies.

6.4.6 Disciplinary Action

The Board shall have the power to dispose of any grievance involving dismissal or disciplinary action by any arrangement that it deems just and equitable.

6.4.7 Expenses

The fees and expenses of the Chair shall be shared equally between the parties. Each party shall be responsible for their costs, fees and expenses of witnesses and those of its Board member.

6.5 General

6.5.1 Recognition of Stewards and Designated Union Representatives

In order to provide an orderly and speedy process for the settling of grievances, the employer acknowledges the rights and duties of Stewards and designated Union Representatives. A Steward or designated Union representative shall represent any employee(s) in investigating, preparing and presenting a grievance in accordance with the grievance process.

6.5.2 Permission to Leave Work

The Union recognizes that Steward **and** designated Union representative **are** employed by the University and that such persons will not leave the workplace during working hours without permission in order to perform duties under this Agreement. Such permission will not be unduly withheld and the Steward **and** designated Union representative will not have their wages deducted nor shall they be hindered, coerced, restrained or interfered with while investigating or presenting a grievance.

All grievances will be dealt with during working hours and no employee or employees, who are representatives of the Union, will suffer loss of pay by reason of time spent in discussing grievances with the representatives of the University or the Union.

6.5.3 Union May Institute Grievances

The Union, its Stewards or designated Union Representatives or its CUPE National Servicing Representative have the right to originate a grievance on behalf of an employee or group of employees or involving interpretation or administration of this Agreement. Such a grievance shall commence at the Human Resources Department. (See Article 6.3)

6.5.4 Facilities for Grievances

The employer shall provide appropriate space for grievance meetings.

6.5.5 Authorized Acting Officials

For the purpose of Article 6, persons duly authorized to act for and on behalf of the officials mentioned in Article 6 during their absence shall similarly be authorized to act in respect to the procedures outlined herein.

6.5.6 Representatives of CUPE

In the discussion of grievances with representatives of the employer, Stewards or designated Union Representatives may at any time be accompanied by a CUPE National Servicing Representative or a Union representative from CUPE **5791**.

6.5.7 Copies of Documents

In the event of a grievance, each party agrees, upon request, to provide the other party with copies of all documents, which the party intends to use in regard to the specific grievance.

ARTICLE 7 -- TYPES OF POSITIONS AND EMPLOYEE PROVISIONS

7.1 Full-Time

A full-time *position* is a position that occurs full days during the regular work week on a continuous basis.

7.2 Regular Part-Time

A regular part-time *position* is a position that regularly occurs on a predictable and scheduled basis and is less than full-time.

7.3 Permanent

A permanent *position* is a position that occurs indefinitely.

A permanent *employee* is an employee who is appointed to a permanent position and has successfully completed the required probationary period.

7.4 Term

A term *position* is a position that occurs for a defined term for a specific project or program or to replace a permanent employee absent for a limited period. The length of a term position may be extended or shortened with two (2) weeks notice in writing contingent on organizational needs and posting requirements. A term position may be either full-time or regular part-time. The employer will include the specifics of a term position on each posting.

A term *employee* is an employee hired for a specific term position.

Term employees shall be entitled to all rights and benefits of this Agreement except where otherwise specified in this Agreement.

Term employees who are appointed for a period of less than one year will not be granted annual leave. Their total remuneration will include an amount of 4/52nds of salary as vacation pay. Term employees appointed for one year or more shall accumulate annual leave entitlements.

Termination of a term employee at the time the term position is discontinued is not subject to the grievance procedure.

The Union recognizes that the Human Rights Code exemption from the Human Rights Commission may be exercised to override the seniority rights of term employees when the employer is filling permanent positions.

Unless undue hardship can be demonstrated by the employer prior to the appointment, in cases where a *permanent employee* is appointed to a term position that does not exceed twenty-four (24) months in duration, at the end of the term appointment, the employee will be returned to his/her previous permanent position. If the employee's former permanent position no longer exists, the provisions in Article 17 will apply.

In cases where a *permanent* employee is appointed to a term position of greater than twenty-four (24) months, the employee foregoes any right to return to his/her previously held permanent position unless the employer and the Union agree otherwise. The employee will retain all other rights of this Agreement except severance pay.

Unless the Union and the employer mutually agree otherwise, upon notification to a permanent employee that the term position will extend beyond twenty-four (24) months and the employee elects to stay in the term position, the employee foregoes any right to return to her/his previously held permanent position. The employee will retain all other rights of this Agreement except severance pay.

When a term employee has been in the same position continuously for more than twenty-four (24) months or in the case of an employee replacing an absent employee due to disability, the employee's status will be changed to permanent.

The employee will then be entitled to all rights of a permanent employee, except the employee will not be entitled to severance pay under Article 17 if the term position is terminated or the employee is not the successful candidate if the position is posted as permanent.

The parties agree a term position shall not be used for altering or changing existing position status without the mutual agreement of the parties.

7.5 Student Worker Positions

7.5.1 Student Worker Positions

Student Worker Positions shall be mutually agreed to by the parties. The agreed to designated positions are Student Library Workers and Student Ambassadors.

7.5.2 Student Library Worker

Student Library Worker positions shall adhere to SCHEDULE D. ALSO, the parties agree to the same general pay increases for the SCHEDULE D employees as those negotiated for other employees.

7.5.3 Summer Student Workers

The Union and the employer hereby agree that the employer may accommodate the employment of summer student workers in the workplace under special project funding or student employment funding opportunities (i.e.: Centennial Student Employment Program or Regina Treaty/Status Indian Services) with the understanding that:

- Summer student worker terms of employment shall fall between May 1 and August 31 of any one year;
- Summer student workers will not perform duties ordinarily performed by Union members;
- 3) Summer student worker positions will not be used to circumvent the creation of potential Union positions; and
- 4) Summer student workers will not be required to pay Union dues.

The employer also agrees to provide to the Union, prior to posting summer student worker positions, position descriptions along with verification that the positions being filled fall under special project funding or a student employment funding opportunity.

7.6 Casual Employees

A casual position is a position that due to its nature does not occur on a predictable and scheduled basis.

A casual employee shall be entitled to all rights and benefits of this Agreement except where specified otherwise.

Casual employees accumulate seniority after 30 days of cumulative service for the purpose of bidding into positions until such time as there is a break in service of greater than three months.

Casual employees after 30 days of cumulative service shall be recalled according to the principles of seniority and qualifications suitable for the position to be filled.

Casual employees will earn seniority according to days worked.

ARTICLE 8 -- SENIORITY

8.1 Seniority Defined

Casual employees are entitled to seniority rights as per Article 7.6.

Term employees accumulate seniority for bidding purposes only. (See Article 7.4)

Term employees who, prior to their appointment to a term position, worked as a casual, will have their casual employment recognized for seniority purposes provided there has been no break in employment of greater than three (3) months.

Seniority rights do not apply during a probation period. Upon successful completion of the probationary period, employees will be credited with seniority from the last date of hire into the bargaining unit. (See Article 9.8)

Seniority for all permanent, term and casual employees is defined as the length of employment from the last date of hire into the bargaining unit subject to any exclusions or interruptions to such as defined in this Agreement. (See Article 8.2)

Permanent employees who, prior to their appointment to a permanent position, and who previously worked on a term basis, will have their employment recognized for seniority purposes provided there has been no break in employment.

Seniority will operate on a bargaining unit wide basis unless specified otherwise in this Agreement. Seniority is a factor in determining promotions, transfers, demotions, bumping, layoff and recall.

8.2 Special Adjustments to Seniority

8.2.1 Temporary Employment Within Another University Bargaining Unit Employees who accept temporary positions in another bargaining unit,

will maintain seniority but will not accumulate it. (See Article 16)

8.2.2 Discharge Without Reinstatement

An employee shall lose seniority rights in the event of discharge without reinstatement.

8.2.3 Resignation

An employee shall lose seniority rights in the event of resignation from the University effective from the termination date. An employee may, without prejudice, withdraw a resignation up to the end of the working day following the day the resignation is submitted.

8.3 Seniority Rosters

The employer will prepare in January of each year and forward to the Union a seniority roster suitable for posting that includes all bargaining unit employees eligible to accumulate seniority.

8.4 Correction of Seniority Roster

On presentation by an employee of proof of error in the roster(s), the employer shall make a correction immediately. The correction will be in the form of a supplementary sheet suitable for posting and forwarded to the Union.

ARTICLE 9 -- VACANCIES, APPOINTMENTS AND PROBATION

9.1 Posting

All vacant bargaining unit positions, whether new or existing, will be **advertised** within the University - Regina Campus, Saskatoon Campus, and Northern Campus and remain open for a period of five (5) working days. A posting via e-mail will be sent to all bargaining unit employees.

As well, all postings, both internal and external, shall be posted on a bulletin board outside the Human Resources Department and in other suitable locations, where appropriate and desirable.

All term positions extending beyond four (4) months shall be posted unless by agreement between the employer and the Union. Any term position that becomes a permanent position shall be posted except for Article 7.4 guidelines.

The postings will include the following information: job title, a description of the position derived from the current position description, required qualifications, location, whether permanent or term, the projected end date of a term position, whether full-time or regular part-time, regular hours of work, pay grade and salary range.

A copy of each posting will be forwarded to the Union.

With agreement from the Union, positions may be advertised outside of the University at the same time as an internal posting with the understanding that bargaining unit applications will be considered before any external application.

9.2 Bidding on Vacant Positions

Employees may bid on posted positions by submitting a resume to the Human Resources Department within the posting period. At the conclusion of the posting period, the employer will provide the Union with a copy of the screening matrix for all internal applicants. Employees may apply for positions on other campuses. Employees on vacation, or approved leaves of absence, may apply by proxy by having a Steward or fellow employee submit an application on their behalf. This application will be given the same consideration as any other application.

The employer will not be responsible for any costs incurred in an employee's efforts to comply with this clause.

9.3 Notice of Results

The employer agrees to make every effort to fill positions and notify applicants as expeditiously as possible following the posting period.

Employees applying for transfer will be advised in writing of the result of their applications by the Employer within two calendar weeks after the vacancy is filled or the competition cancelled.

9.4 Basis for Selection

See Articles 7 and 8 for seniority exclusions and limitations related to the following provisions.

Any posted vacancy shall be filled with preference to bargaining unit employees with the required qualifications as stated in the job posting, ability sufficient to perform the position and the greatest seniority. Any outstanding performance issues/disciplinary actions documented on the employee's official file or destined for the employee's official file will be a consideration in determining "required qualifications" and "ability sufficient". (See Article 4.6)

If there are no fully qualified CUPE bargaining unit applicants, the employer will consider, in order of seniority, the applications of employees who are close to possessing the required qualifications before considering applicants from outside of the CUPE bargaining unit. If an appointment is made of an applicant who does not possess all of the required qualifications of the position, the employer may, as a condition of appointment, require that the applicant obtain the qualification(s) within a specified time limit.

9.5 Provisional Period

A permanent employee who is appointed, placed, recalled, bumps or accepts a transfer to another permanent position within the bargaining unit shall be appointed provisionally for three (3) months, during which time performance will be appraised. Such appraisals will be discussed with the employee.

An employee who successfully completes the provisional period shall be declared permanent in the position.

9.5.1 Extension

At the discretion of the Human Resources Department, the provisional period may be extended by the cumulative length of any period(s) of absence from work for more than five consecutive days. In the event this happens, the employee will be notified in writing at least two (2) weeks prior to the extension.

Notwithstanding the above, the provisional period may also be extended by a period of up to three months if mutually agreed to in writing by both parties to this Agreement.

9.5.2 Reversion

At the end of the provisional period, if the employee has not performed satisfactorily, or at any time during the period when it becomes clearly evident that the employee is incapable of handling the job, the employee shall be returned to the former status (either layoff or position and salary) subject to any performance increment, payable no sooner than the reversion date, which normally would have been received had the employee remained in that position.

During the provisional period, the employee may revert voluntarily to the employee's former status (either layoff or position and salary) subject to any performance increment, payable no sooner than the reversion date, which normally would have been received had the employee remained in that position.

Each employee can exercise this voluntary return to former status right only once as a bargaining unit employee. The employee's offer letter from the employer shall reference this return to former status right.

During the provisional period of appointment, the CUPE employee's previously held position may be filled on a term basis.

In the event of a reversion, the vacated position will not be re-posted but rather, the selection and hiring process will continue with the remaining list of original applicants, as if no appointment had been made prior to the reversion.

If such reversion displaces an employee who has completed a probationary or provisional period, the displaced employee shall have the right to revert or to be laid off, as appropriate.

In the event of reversion to a position that has been eliminated under Article 17, the employee will not return to his/her previous position but rather, will be dealt with under the applicable provisions of Article 17.

9.6 On-the-Job Training

At the written request of an employee and with the written agreement of their supervisor, arrangements may be made for on-the-job training conditional on no disruption of the performance of the duties of any positions affected. Length of service in the work unit will be one of the factors considered in scheduling the training. This may be on the employer's time or arrangements can be made to allow access to the facilities of the workplace on the employee's time.

9.7 Probationary Period

9.7.1 Extension Due to Absence

At the discretion of the Human Resources Department, the probationary period may be extended by the cumulative length of any period(s) of absence from work for more than five consecutive days. In the event this happens, the employee will be notified prior to the extension.

9.7.2 Extension By Agreement

Notwithstanding the above, this probationary period may also be extended by a period of up to three months if mutually agreed to by both parties to this agreement.

9.7.3 Termination During Probationary Period

During their probationary period, employees will be advised of expectations regarding standards of performance. Employees will also be advised in writing of shortcomings in order to correct deficiencies. Should the employer decide to terminate the employee, the employee will be given the reasons in writing with a copy to the Union prior to the employee's termination and provided an opportunity to respond.

Seven days notice shall be given if the incumbent has been in the position for less than fifteen (15) weeks. If the incumbent has been in the position for fifteen (15) weeks or longer, two (2) weeks notice will be given. A copy of the notice shall be provided simultaneously to the Union.

Termination of an employee during the probation period is subject to the grievance procedure.

A Steward or designated Union Representative shall be present during all of the above processes.

9.8 Length of Probationary Periods

All appointees to a permanent position, who are not already permanent employees, shall be on probation for a period of six months from the date of commencing duties in the permanent position.

Unless the employee is already a permanent employee, a term appointment to a term position shall be probationary for a period of six months from the date of commencing duties in the position. (See Article 9.5)

During the probationary period, the employee shall be entitled to all rights and benefits of this Agreement except where specified otherwise in this Agreement.

ARTICLE 10 -- WORKLOAD ISSUES

10.1 Requests for Review

When an employee feels the workload, when balanced over a reasonable period of time, is significantly greater than it should be, the employee may discuss the matter with his/her supervisor in the presence of a Steward or **designated Union Representative**. If the matter is not resolved, the employee may request the Union to address the issue at a joint Union-Management Committee meeting at which the employee and the supervisor shall be present.

10.2 Dispute Resolution

If a satisfactory resolution cannot be reached in meeting, the employee may grieve under the provisions of Article 6.3.

ARTICLE 11 - HOURS OF WORK AND SPECIAL PAY PROVISIONS

11.1 Daily Hours of Work

Article 11.1 (a) and (b) encompasses all employees unless otherwise stated in this Agreement.

The employer agrees to negotiate with the Union, as the need arises and for all bargaining unit positions, further changes to provisions under Article 11.1: Daily Hours of Work.

- (a) Seven (7) hours shall constitute an ordinary days work and thirty-five (35) hours a week prevails. Daily hours of work shall fall between the hours of 8:00 a.m. and 6:00 p.m., Monday through Friday inclusive. The daily hours shall be worked within an eight-hour period.
- (b) Employees who work full days (7 hours) will be permitted two fifteenminute rest breaks or one thirty-minute rest break per day. Employees

who work half days (3.5 hours) are entitled to one fifteen-minute rest break. Unused rest breaks may not be used to alter hours of work in any day. Breaks will be arranged to maintain at least minimal service in any **department**, where possible.

(c) Notwithstanding Article 11.1 (a) and (b),

Daily Hours of Work: Plant, Property and Maintenance Employees

- The normal hours of work for Plant, Property and Maintenance employees shall be seven (7) hours per day and thirty-five (35) hours per week, between the hours of 6:00 a.m. and 11:00 p.m., Monday through Friday. The daily hours of work shall fall between the hours of 6:00 a.m. and 6:00 p.m. and shall be worked within an eleven (11) hour period.
- An 'hours of work' schedule shall be posted in a place accessible to the employees. The employer shall provide Operator/Caretakers and the Union with fourteen (14) days written notice of any change to the daily hours of work.

Operator/Caretakers shall have the option of trading days providing another employee is willing to trade and providing the supervisor is given two (2) days written notice and approves the trade.

- iii) An Operator/Caretaker shall receive the benefit of Article 11.3.8: Unsocial Hours of Work Premium if the employee is scheduled to work outside the hours of 6:00 a.m. to 6:00 p.m. on any day of the week.
- iv) If an Operator/Caretaker's scheduled work day falls on a named holiday as per Article 20.1: Named Holidays, the employee shall be paid double time and one-half for all hours worked.
- v) If an Operator/Caretaker's scheduled day off falls on a named holiday as per Article 20.1: Named Holidays, the employee shall be paid for seven (7) hours of work as though the employee was actively at work on that named holiday.
- d) Notwithstanding Article 11.1 (a) and (b):

Daily Hours of Work: Other Plant, Property and Maintenance Employees

 Seven (7) hours shall constitute an ordinary days work and thirty-five (35) hours a week prevails. Daily hours of work for Plant, Property and Maintenance employees shall fall between the hours of 6:00 a.m. and 6:00 p.m., Monday through Friday inclusive. The daily hours shall be worked within an eight-hour period. Article 11.3.8: Unsocial Hours of Work Premium shall apply.

 The employer shall provide Plant, Property and Maintenance employees and the Union with fourteen (14) days written notice of any change to the daily hours of work.

11.2 Part-Time Employees

Except in the case of emergencies, employees who are employed to work fewer than the stated regular hours of work may not be required to work additional hours without mutual agreement.

11.3 Accommodation of Employees Who Are Students

The employer will make every effort to accommodate the academic schedule of employees who are full-time students when assigning work schedules. In the event of two such employees claiming the same work schedule, the employee with the greatest working service at the University will be given preference. (See Article 7.5 and SCHEDULE D)

11.4 Overtime

All overtime arrangements must have prior written authorization from the appropriate supervisor except in emergency situations when there is no supervisor reasonably accessible to authorize such overtime.

11.4.1 Overtime Rate

Employees shall be paid, at their current hourly rate as determined by the instructions in Schedule A of this Agreement, time and one-half for all time worked in excess of the stated regular hours of work.

11.4.2 Time Off in Lieu of Overtime

Any employee required to work overtime shall receive pay for this time unless the employee and the supervisor mutually agree in writing to time off at the appropriate overtime rate in lieu of pay.

11.4.3 Accumulation of Time Off in Lieu of Overtime

Employees may accumulate overtime compensation from each incidence of overtime for up to a twelve month period in order to provide time off with pay. Such time must be taken within three months of the end of the twelve-month period at a time mutually agreed in writing between the supervisor and the employee.

11.4.4 Part-Time Overtime

Employees who work less than full days will not receive overtime until their work exceeds the stated regular hours of work as specified in Article 11.1.

11.4.5 Overtime Voluntary

Overtime shall be voluntary except in emergencies as defined in **The Saskatchewan Employment Act**. In the event there is a difference of opinion between an employee and his/her supervisor as to what constitutes an emergency, the final decision shall be that of the Director of Human Resources.

11.4.6 Employer Directed Overtime

When an employee is directed by the employer or supervisor to attend a retreat or other event outside of the employee's regular hours of work and / or overnight, the University will pay:

- (a) the cost of the employee's transportation, food and lodging expenses as per FNUiv policy.
- (b) If at all possible, travel time will be scheduled during regular work hours and, if otherwise, will be paid at the overtime rate.

11.4.7 Call Out

11.4.7.1 Overtime

Overtime shall be voluntary. If an employee has left the place of work and is called back for overtime work, the employee shall be paid a minimum of three hours at overtime rates.

If no employee(s) volunteer, in an emergency situation, the employer can, using reverse seniority, direct employee(s) to return to work.

11.4.7.2 Scheduled Employees

Employees working according to a posted schedule who are called back to work shall be paid a minimum of three hours at overtime rates.

11.4.7.3 Phone Calls to Employees

Phone calls made by a supervisor from the workplace to an employee's personal/home phone number for work consultation purposes shall be paid a minimum of three (3) hours at their hourly wage.

11.5 Unsocial Hours of Work Premium

Unless otherwise stated in this Agreement, for time actually worked, an employee will receive an additional **one dollar (\$1.00)** per hour for each hour or part of an hour of regularly scheduled work outside the hours of 8:00 a.m. to 6:00 p.m. Monday through Friday and for all hours worked on Saturday and

Sunday. If the majority of an employee's hours of work fall outside the hours specified above, the premium will be paid for the entire work day.

Unsocial hours of work premiums do not apply to overtime.

11.6 Provision of Meal

Where an employee is required to work continuously beyond normal quitting time in excess of two hours, or is required to return to work for overtime purposes for a period in excess of four hours, the employer agrees to provide a suitable meal.

11.7 Childcare Expenses

When an employee is required by a supervisor of the employer to be away from home overnight at the request of the supervisor or employer and when the employee has to hire someone to care for his/her children during this absence, childcare costs will be compensated by the employer to a maximum of \$50.00 per day per child under the age of sixteen for childcare.

A signed receipt from the care provider must accompany the expense claim.

11.8 Provision of Tools

The employer shall supply all tools and equipment required by the employee in the performance of the employee's duties. Tools will be replaced when the employee produces the worn or broken tool. The employee shall return all tools and equipment upon leaving the employment of the University.

11.9 Bilingual Bonus

An employee shall receive a bilingual bonus of four (4) percent per month providing the position requires, on a frequent and continuing basis, fluency in speaking, reading or writing in a language other than English, and the position is designated a bilingual position by the Human Resources Department.

Employees may apply in writing to the Human Resources Department, with a copy to the employee's supervisor, requesting a bilingual designation. Detailed justification supporting the request must be included.

11.10 Clothing and Protective Clothing

Clothing worn by employees at work will be neat and appropriate to the employee's duties.

Adequate protective clothing will be provided by the employer when the duties performed by an employee might result in the employee's clothing being soiled, destroyed or rendered unfit for further use.

Where employees are required to wear specific clothing or from a collection of specific clothing selected by the employer, the employer will pay the cost of the clothing and any alterations required.

11.11 Receipted Expenses

Employees will be reimbursed for reasonable receipted expenses not addressed in this Agreement while on University business.

11.12 Employment Travel Expense

The employer will reimburse employees for travel expenses incurred in performing their duties provided approval is received prior to the travel occurring.

Employees shall be reimbursed for employment travel expenses in accordance with University policy, including any additional expenses referenced in Article 11.

11.13 Itemized Statement

The employer will provide, on each pay day, to each employee, an itemized statement of wages showing the month, hours, rates, deductions, etc. Upon request, personal material will be provided in sealed envelopes.

11.14 Pay for Temporary Performance of Higher Level Duties

11.14.1 Within the Bargaining Unit

When an employee is expected to perform temporarily the duties of a higher paid position or is assigned those duties in writing by the employee's supervisor, the CUPE salary of the employee shall be adjusted from the date of reassignment to the greater of:

- 1) the step in the pay band of the higher paid classification which is next higher than the employee's current salary rate; or
- 2) a premium of 6% of the employee's current salary rate; or

The employee will be paid at the higher rate for the duration of the assignment.

The employee has the right to refuse to perform higher-level duties.

11.14.2 Outside the Bargaining Unit

Prior to an employee performing duties outside the bargaining unit, the employer, the union and the employee shall meet to determine the terms of the assignment.

The employee has the right, at any time, to refuse to perform higher-level duties.

11.15 Administrative Errors

Administrative errors made relative to an employee's salary or benefit plans will be adjusted in such a way as not to prejudice the rights of the employee. In any instance where recovery is required, the employer and Union will meet with the employee to develop a reasonable repayment schedule. Failure to agree will result in the employer implementing a reasonable repayment schedule.

ARTICLE 12 -- PERFORMANCE EVALUATION AND INCREMENTS

The parties agree the purpose of performance evaluations is to provide the employee with an honest and fair assessment of performance. All persons involved in the review will undertake their roles seriously and with integrity, ensuring that statements, both verbal and written, refer to aspects of performance, are fair commentary and are based upon appropriate evaluation and evidence.

Changes to the Performance Evaluation Form will be discussed with the Union prior to alterations.

12.1 Process and Responsibilities

Approximately thirty (30) days prior to the employee's annual performance increment anniversary date in his/her current position, the supervisor may initiate a performance evaluation by forwarding a <u>completed</u> Performance Evaluation Form to an employee under his/her direction.

The employee will have ten (10) working days to review and prepare a response, at their discretion, as an attachment to the Performance Evaluation Form before returning the form to their supervisor. The employee may request the assistance of a Steward or designated Union Representative in preparing a response.

The employee may have a Steward **or designated Union Representative** present as an observer during the performance evaluation discussion.

The employee shall sign the performance evaluation as proof the employee has read it; such signature shall not necessarily constitute agreement with the contents of the evaluation.

The supervisor is ultimately responsible for completion of the performance evaluation process, which includes discussing the performance evaluation with the employee, obtaining signatures and forwarding the Performance Evaluation Form to the Director of Human Resources where it will be placed in the employee's official file. The employer will forfeit the right to withhold an increment if the supervisor fails to complete the evaluation process by the employee's anniversary date. Increments may be withheld only when the performance evaluation documents unsatisfactory performance of duties and the employer has made a reasonable effort to coach the employee about his/her work performance.

12.2 Performance Evaluation Grievances

The supervisor shall provide a copy of the completed Performance Evaluation Form to the employee who shall have five working days to review and prepare a response at their discretion. The employee may request the assistance of a Steward in preparing a response.

The employee may include a response to the performance evaluation as an attachment to the Performance Evaluation Form before returning the form to their supervisor.

If an employee does not agree with the contents or outcome of their employee performance evaluation, the employee shall have access to the grievance and arbitration process.

12.3 Changes to Anniversary Date

Performance increments are effective on an employee's annual anniversary date and will be paid no later than two pay periods after the anniversary date.

When an employee is reclassified under Article 37 to a higher pay grade or appointed to a new higher pay grade position, the date of the position change will become the employee's new anniversary date.

When an employee is laterally transferred to a position with the same pay grade or to a lower pay grade, an employee's anniversary date does not change.

12.4 Working Service

Working service includes annual leave and leave for jury duty. Working service excludes the cumulative total of all other absences greater than four consecutive weeks during the annual performance review period, **except for union leave**.

Employees with 52 full weeks of working service from their anniversary date are eligible for a full performance increment. Employees with less than 52 full weeks of working service from their anniversary date shall have their increment anniversary date set back accordingly.

Pay range ceilings apply.

An employee who takes parental/maternity leave will not be penalized for his/her absence from work. The employee will return to his/her employment status held at the commencement of the leave and the performance increment date deferred to account for a full year of working service. In addition, any general salary increases to the CUPE salary grid will be applied to his/her salary upon return.

12.5 Financial Exigency

The parties agree that performance increments are paid subject to satisfactory performance. If increments are withheld, any adjustments to salary, or compensation of any kind for individuals employed or funded by the University, directly or indirectly, shall be made on a basis that is fair, equitable and consistent in approach.

12.6 Preferential Treatment

In no case shall any increase be provided to a person not covered by this Agreement that exceeds the average increase provided to persons within this Agreement.

ARTICLE 13 -- DISCIPLINE

13.1 Principle of Innocence and Burden of Proof

Both parties agree that an employee is considered innocent until proven guilty. In case of discharge or discipline, the burden of proof of just cause shall rest with the employer. Evidence presented shall pertain only to the grounds stated in the discharge or discipline notice to the employee.

The employer endorses the concept of progressive discipline in situations of poor performance. While the employer reserves the right to use any disciplinary action deemed appropriate, regardless of the order of the following clauses, the parties agree that, where possible, an employee should receive a verbal warning of inappropriate behaviour before written reprimands or other types of disciplinary action are taken.

Verbal reprimands shall not be documented in the employee's official file, are not grievable and shall be delivered in the presence of a Steward **or designated Union Representative**.

13.2 Written Reprimand

If an employee is formally reprimanded concerning unacceptable conduct or performance, it will be done with Steward **or designated Union Representative** present. A written reprimand including particulars of the work performance or behaviour, which resulted in the reprimand, shall be forwarded to the employee within ten working days of the reprimand discussion with copies to the Union and the Human Resources Department. A copy of the written reprimand will be placed on the employee's official file. An employee may respond in writing to the reprimand, and such response will become part of the employee's official file. Written reprimands will be removed from the employee's file after two years of subsequent employment during which no additional documented disciplinary action is taken.

Written reprimands are grievable by the Union.

13.3 Written Warning

The Union shall be notified of the issuing of a written warning forty-eight (48) hours (two working days) in advance of the warning being given to the employee. The written warning shall be given to the employee with a Steward **or designated Union Representative** present. Copies of the written warning will be forwarded to the Union and the Human Resources Department. A copy of the written warning will be placed on the employee's official file. An employee may respond in writing to the reprimand, and such response will become part of the employee's official file.

Written warnings will be removed from the employee's file after two years of subsequent employment during which no additional documented disciplinary action is taken.

Written warnings are grievable by the Union.

13.4 Suspension

The employer reserves the right to suspend an employee for just cause, with or without pay depending on the severity of the infraction, for a period of up to one month. The suspension shall be given to the employee with a Shop Steward or designated Union Representative present. Confirmation of the action to be taken by the employer will be conveyed to the employee in writing by the Human Resources Department as soon as possible, with a copy of the suspension letter simultaneously sent to the Union. If the Union considers the action unjustified, it shall have from the date the suspension begins or the notice is received by the Union, whichever is later, fourteen calendar days in which to file a grievance.

Suspension letters will be removed from the employee's file after two years of subsequent employment during which no additional documented disciplinary action is taken.

13.5 Dismissal

The employer reserves the right to dismiss any employee for just cause. The dismissal shall be done in the presence of a Shop Steward or designated Union Representative. Written confirmation of the action taken will be conveyed to the employee in writing by the Human Resources Department with a copy sent simultaneously to the Union. The employee shall be suspended for

fourteen (14) calendar days with pay during which time the Union shall have the opportunity to investigate the circumstances and state its case.

Unless a grievance is presented to the employer within the fourteen-day period following the receipt of a copy of a dismissal letter, the employee will be dismissed. If no just cause for dismissal can be proved, the employee shall be reinstated without loss of pay or with a lesser disciplinary action as the parties may agree is appropriate to the circumstances.

13.6 Benefits During Suspension

An employee's benefit plans will continue in full effect during any suspension.

ARTICLE 14 -- RESIGNATION

An employee who resigns is expected to give at least one month's notice in writing to the appropriate supervisor with a copy to the Human Resources Department. The length of the notice period may be changed or waived by mutual agreement between the employee and the appropriate supervisor. (See Article 8.2.3)

ARTICLE 15 -- RETIREMENT

The pension plan referenced throughout Article 15 refers to the Group Retirement Services pension plan addressed in Article 23.

15.1 Retirement Age and Benefit Coverage

Normal retirement is the first of the month coinciding with or following the employee's sixty-fifth birthday. By giving three months notice, an employee may retire any time in the ten years **preceding** the normal retirement date.

An employee may continue working and defer receiving pension payments. After age 65, continued employment will be conditional on satisfactory annual performance evaluations. Contributions to the pension plan from both employee and employer will be made for employees working after the normal retirement date until the employee retires or until pension payments begin.

15.2 Special Arrangements

Without prejudice, the employer may propose a plan to an employee who is not eligible for retirement whereby the employee's employment may be discontinued in accordance with a special arrangement, including a suitable financial settlement. A member may submit a request to the employer to initiate such an arrangement.

The employer will keep the Union and the employee who is eligible for such a special arrangement apprised of the basic terms of the arrangement with the

understanding that there may be some flexibility in order to provide for the particular needs of the individual.

Without prejudice, the employer may offer an early retirement package to an individual employee or to a group of employees who are eligible to take early retirement and who meet certain conditions as outlined in the University's pension plan.

Whenever the employer proposes a plan for discontinuance of employment or assisted early retirement, the employer shall inform the Union, in writing, of the terms of any such formal offer two full days prior to making the formal written offer to the employee(s).

Any offer made in writing shall remain open for a minimum of thirty (30) days. Employees have the right to have a Union representative participate in any discussion that may take place between the employer and the employee after the formal written offer is made to the employee.

Any agreement or arrangement concluded must be in writing. The arrangement will not take effect until the President, the affected employee(s) and the Union have signed the agreement.

ARTICLE 16 - ENTRY AND RE-ENTRY FROM EXCLUDED POSITIONS

An employee who is appointed to a term position excluded from the bargaining unit shall cease membership and discontinue paying dues for the duration of the appointment provided the appointment is for more than one (1) calendar month At the termination of the appointment to an excluded position, the employee shall automatically revert to his/her former position, become eligible for CUPE membership and commence paying dues. The employee shall have all the rights and privileges which are possible at the time, such as general salary increases and performance increments, restored effective the date of return to the bargaining unit, as if having been a CUPE member continuously throughout the period of appointment to the excluded position. Seniority will be adjusted accordingly provided the appointment is for more than one (1) calendar month.

Unless the Union and the University mutually agree otherwise, the term position will not extend beyond 24 months.

A University employee from outside the bargaining unit who is appointed to a position in the bargaining unit and who had previous continuous employment with the University, shall have all the rights and privileges which are possible at the time, such as annual leave entitlements except for the probationary requirement and seniority, as if having been a CUPE member continuously

throughout their University employment. When entitled, the employee's seniority shall date from the time the employee enters the bargaining unit.

ARTICLE 17 - STAFF REDUCTION

17.1 Determination

Should the employer determine a need to reduce the hours of work in or eliminate any positions based on financial necessity, a reduced need due to restructuring, elimination or transfer of any University-sponsored programs or services, or as a result of a reduced need due to technological change, the affected permanent incumbent will be dealt with in the manner outlined below. (See Article 17.2.3)

Term and casual employees are excluded from these provisions.

For an employee in a term position of indefinite length, the notice period for the purpose of Article 17.2.2 only, is deemed to commence with the notice of a specific termination date in that position.

17.2 Notice

17.2.1 Notice To the Union

The employer shall provide as much notice as possible to the Union; however, notwithstanding, the employer shall notify the Union in writing at least two (2) working days prior to notifying the affected employee(s) of the reduction in or elimination of their position(s). The following details will be provided:

- 1) the positions (title and location) that will be affected;
- 2) to what extent each position will be affected;
- 3) the reason for the intended reduction or elimination; and
- 4) the intended date of employee notification.

17.2.2 Notice To the Employee

The employer will provide the affected employee(s) with notice as required in accordance with the Labour Standards Act of Saskatchewan. In no instance will the employee(s) be given less than thirty calendar days notice.

The employer will notify each employee and the Union in writing:

1) of the reduced hours or elimination of the position;

- of the reason for the reduced hours or elimination of the position; and
- of whether the employer intends for the employee to remain employed in his/her position with the University during the notice period or that the employee will receive pay in lieu of the notice period prorated.

If the employee qualifies, the employer will include the details of an offer of assisted early retirement as part of the notification package to the employee. (See Article 17.3)

Within five working days of the reduction notification to the employee, the employer shall supply the employee with the following information for bumping consideration:

- 1) a CUPE seniority roster;
- position description or job posting, whichever is most recent, for positions that qualify and have an incumbent with less seniority than the bumping employee; and
- 3) current work schedule of above-mentioned positions.

Throughout the notice period the employer will provide the affected employee(s) with reasonable information such as position descriptions and/or job postings toward informed decision-making regarding other suitable employment opportunities at the University.

In addition, the Human Resources Department will offer counseling services. The employee may be referred to other sources for additional counseling based on need. A union representative, Elder or other support person (excluding a parent, spouse, son or daughter) may be present at meetings with Human Resources.

17.2.3 Technological Change Consultation

In an attempt to keep employees apprised of current and anticipated modifications to the day-to-day work techniques of various classes of employees, the employer and the Union agree to meet from time to time as necessary. In addition, when the Human Resources Department becomes aware of impending significant technological change at the University, it will undertake to call such a meeting. In cases of technological change, which directly affects conditions of employment, the employer and the Union agree to enter into consultation at the request of either party.

Consultation may include such things as:

- the nature of change to be introduced;
- timing of such; reassignment of duties;
- effects on terms and conditions of employment;
- plans for retraining relative to existing employees adapting to new equipment or work methods;
- establishment of a rate of pay to be provided during training and arrangements for the costs of materials and/or tuition;
- arrangements for assessment of an employee's suitability for training;
- arrangements for periodic assessment of an employee's progress while in training.

17.3 Assisted Early Retirement

17.3.1 Assisted Early Retirement Offer

If the employee qualifies, the employer will include the written details of an offer of assisted early retirement as part of the notification package to the employee. The employer will inform the Union of the terms of the offer. (See Article 15)

17.3.2 Notification by the Employee

The affected employee may elect to accept the early retirement offer by notifying the Human Resources Department, in writing, within fourteen calendar days of written notice from the employer. Upon election of this option, the employee will forego any other options outlined in Article 17.

An employee opting to accept an offer of assisted early retirement will cease being an employee on the first calendar day after his/her notice period from the employer expires, foregoing any further rights under this Agreement unless otherwise specified in this Agreement. The first calendar day after his/her notice period expires shall also be deemed the employee's retirement date.

17.4 Redeployment and/or Retraining

17.4.1 Redeployment and/or Retraining Offer

The employer will make every attempt to provide employees with offers of redeployment or retraining. Such training may be for an existing permanent position, a new permanent position or to supplement an employee's skills.

The employer will include written details of an offer of redeployment and/or retraining as part of the notification package to the employee. The employer will notify the Union of the terms of the offer prior to presentment to the employee. The affected employee may elect to accept the offer of redeployment and/or retraining by notifying the Human Resources Department, in writing, within fourteen (14) calendar days of written notice from the employer.

17.4.2 Provisional Period

A permanent employee who is transferred shall be appointed provisionally for three months, during which time performance will be appraised. Such appraisals will be discussed with the employee in the presence of a Steward, if the employee so chooses.

An employee who successfully completes the provisional period shall be declared permanent in the position and will forego any other options outlined in Article 17.

At the discretion of the Human Resources Department, the provisional period may be extended by the cumulative length of any period(s) of absence from work for more than five consecutive days. In the event this happens, the employee will be notified in writing at least two (2) weeks prior to the extension.

Notwithstanding the above, the provisional period may also be extended by a period of up to three months if mutually agreed to in writing by both parties to this Agreement.

At the end of the provisional period, if the employee has not performed satisfactorily, the employee shall be provided the balance of their options under Article 17.

17.5 Termination and Severance

17.5.1 Notification by the Employee

Subsequent to the notification of reduced hours in or the elimination of an employee's position, the employee may elect to terminate employment and receive severance pay by notifying the Human Resources Department within fourteen calendar days of written notice from the employer. Upon election of this option, the employee will forego any other options outlined in Article 17.

17.5.2 Termination Date

The employee will cease being an employee on the first calendar day after his/her notice period from the employer expires, foregoing any further rights under this Agreement.

17.5.3 Calculation of Severance Pay

Severance pay shall be calculated at a rate of one month's pay at the employee's current rate of pay for each full year of working service. Partial years of employment shall be prorated for this calculation. Regardless of years of working service, the maximum amount payable by way of severance shall be limited to twelve (12) months pay and payable to the employee on the last day of their notice period. (See Article 17.7.8)

17.6 Bumping

17.6.1 Notification by the Employee

If the employee does not elect early retirement (where applicable), redeployment and/or retraining as offered, layoff and recall or termination and severance, the employee may elect to exercise bumping rights by providing written notice to the Human Resources Department within fourteen (14) calendar days of written notice from the employer, including notice of which position the employee wishes to bump. Upon election of this option, the employee will forego any other options outlined in Article 17.

17.6.2 Bumping Rights

Only permanent positions will be considered for bumping. An employee may bump another Union member with less seniority. For the purposes of this paragraph, the Regina Campus, the Saskatoon Campus, the Northern Campus and NSDT shall each be treated as one campus.

Employees choosing to bump to another campus will be responsible for moving expenses.

Bumping rights shall accrue as follows:

- First into a job title where the position is similar to the employee's current position (same pay grade) provided that the employee meets the minimum qualifications for the job.
- Second, into any position at the same or lower pay level provided that the employee meets the minimum qualifications for the job as it was last posted or evaluated, as the case may be, except when the employer can demonstrate to the Union a bona fide change in the position requirements.
- Employees who bump into a different job title and pay grade shall be placed on a provisional period.
- 4) The employee who is bumped shall immediately have access to the provisions of Article 17 except that the requirement for notice, Article 17.2, will not apply. If there is advance notice of a bump occurring, the affected employee shall be dealt with in accordance with Article 17.4, and all subsequent clauses in this article to the extent that time and opportunity permit.

The employee will be placed into the new position on the first working day following the notification period from the employer.

Notwithstanding the provisions of Article 17.6, at the request of the employee, the employer and the Union may agree on another arrangement, without prejudice, for a bumping situation. The employee must submit a letter requesting special consideration to the Human Resources Department prior to the mid-point of the employee's notice period from the employer.

17.7 Layoff and Recall

An employee who does not notify the Human Resources Department of their intention within the notification periods as outlined in Article 17, will be laid off subject to recall.

Laid off employees shall be advised of re-employment options by the employer in the presence of a Union representative.

17.7.1 Applications for Re-Employment

Laid off employees shall provide the Human Resource Department with a current resume within fourteen (14) calendar days of the start date of their lay off. When a new bargaining unit position is created or a position becomes vacant, the employer will automatically enter the laid off employee's name with the names of other applicants from within the bargaining unit. The position will be filled in accordance with Articles 9.2.2 and 9.4.

17.7.2 Election to Work Casual or Term

The employee may elect to work in available casual or term positions for which they have the qualifications and ability to perform the duties required for the position, without prejudicing their right to recall. Such casual or term employment shall suspend the employee's cumulative layoff for the duration of the employment. Upon expiry of the term position or if the employee fails the provisional period, the employee will be returned to cumulative layoff status.

17.7.3 Reporting for Duty

If a laid off employee is successful in their application to a posted position in Article 17.7.1 above, they shall report for duty on a mutually agreed date. If there is no mutually agreed date, the employee shall report for duty as specified in the appointment letter sent by registered mail to the employee's last known address. Failure to report for duty as specified will automatically cancel the awarding of the position to the employee. The employee will remain on layoff status. The employer will then award the position to the next qualified applicant. If the next or subsequent successful applicants are also on layoff, this clause will continue to apply until the position has been filled.

17.7.4 Decline of Offer

If a laid off employee is awarded a posted position through the application of this article, they shall be allowed one occasion in which to decline a position or fail to respond to an appointment. If the employee fails to respond to a second appointment or declines the awarding of the position on a second occasion by failing to report for duty within the ten calendar day period, the employee's employment shall be terminated unless the failure(s) to return to work have been because of circumstances beyond the employee's control.

17.7.5 Address Advice

It shall be the responsibility of all employees, including those laid off, to keep the Human Resources Department advised in writing of any changes to their address and/or telephone number. The Union will not hold the employer responsible for their inability to contact an employee due to an outdated address or telephone information.

17.7.6 Provisional Period on Recall

Employees who are recalled from layoff to a permanent or term position shall be considered to be on a provisional period in accordance with Article 9.5. If their performance is deemed to be unsatisfactory within this period of time, they will be returned to layoff and not recalled to the same position. The time spent in the position in question will not be included in the calculation of cumulative layoff.

17.7.7 Salary Administration

An employee who is recalled to or bumps into a position in a lower pay grade than their previous position, shall be offered the salary within the new pay grade that takes into account the employee's previous working service within the bargaining unit.

17.7.8 Severance After Layoff

When a permanent employee has been laid off for a cumulative period of twelve months without having successfully regained permanent employment, or the employee chooses to give up their recall rights, the employee will be considered severed and will be issued severance pay.

17.7.9 Grievances

Grievances concerning layoff and recall shall be initiated at $\mbox{Stage 2}$ of the grievance process.

17.7.10 Benefits During Layoff

The laid off employee may make arrangements with the Human Resources Department to continue paying the employee portion of the premiums for supplementary health care and/or dental care throughout the first six (6) months of the cumulative layoff period. The employer will continue to pay the employer's portion of those premiums selected by the employee. Should the employee choose to give up their recall rights, premium payments will be discontinued except where the employee is 55 or will reach the age of 55 in the year of layoff or during the year after the year of layoff. In these instances, Article 15.1 will apply.

ARTICLE 18 -- CLEARANCE UPON TERMINATION

Upon termination of employment, the final salary cheque will be issued within six (6) days of the last day on payroll, or as soon thereafter as all financial and material obligations of the employee to the employer are satisfied. Such obligations may include, but are not limited to, the return of all keys, identification cards, library cards, audio-visual and other equipment, and reimbursement for goods, services and outstanding travel advances.

ARTICLE 19 -- PAYMENT TO ESTATES OF DECEASED EMPLOYEES

When an employee has died, the employer will pay to the beneficiary as designated and specified by the employee in writing, the employee's salary up to the date of death in full, minus normal deductions. In the event of failure to designate a beneficiary, the employer shall make payment to the estate of the deceased employee.

ARTICLE 20 -- HOLIDAYS

20.1 Named Holidays

Notwithstanding Article 1.3, the employer recognizes the following paid holidays:

New Year's Day Indian Government Day Family Day Good Friday Victoria Day First Nations Solidarity Day Canada Day Saskatchewan Day Labour Day Thanksgiving Day Remembrance Day Provincial First Nations Unity Day Christmas Day Boxing Day Treaty Day (on the treaty day of the employee's band)

The Christmas Break will be the period designated by the University of Regina and will be with pay including December 25, 26 and January 1.

The parties agree that the University may, in order to keep its calendar consistent with the University of Regina's calendar, move specific First Nations holidays as necessary. The employer will post a paid holiday schedule for the current year in January of each year.

Employees scheduled to work during the Christmas break will receive pay at time and one half at their current hourly rate as determined by instructions in Schedule A of this Agreement, for all hours worked along with any other holiday pay provisions as outlined in this Agreement.

20.2 Holiday Pay

Every full-time worker will receive one days pay for each of the above named holidays.

Each less than full-time worker will receive holiday pay based on the employee's total paid hours (exclusive of overtime) in the four weeks preceding the holiday divided by 20.

20.3 Observation of Named Holidays

Excluding the First Nations holidays, if the holiday falls on a Saturday or Sunday, the holiday will be observed on the following Monday. If both the Saturday and Sunday are holidays, the holidays will be observed on the following Monday and Tuesday.

20.4 Compensation for Work on a Holiday

An employee who works on a holiday at the request of the employer and is not approved to take another mutually agreed day off in lieu, shall be paid time and one-half for all hours worked in addition to the holiday provided in Article 20.2.

20.5 Closure Due to Inclement Weather or Other Emergency

When the Employer issues a notice that University offices are closed due to inclement weather or other emergency, employees will not be deducted pay because of the notice. Employees who are required to report to work will receive the equivalent of straight time off and their regular pay.

The Employer will post notices of closure on the University web site as well as by e-mail to all employees at the location(s) affected.

ARTICLE 21 - ANNUAL LEAVE

21.1 Accumulation

The following provisions accrue to full-time and regular part-time employees except casual and term employees appointed for a period of less than one consecutive year of service with the employer. Annual leave entitlements shall accumulate according to the number of consecutive years of service with the employer as follows:

Less than five years of service

1.66 days per month

Five to ten years of service	2.08 days per month	
Over ten years of service	2.50 days per month	

Annual leave entitlements shall be prorated for all casual and term employees who are appointed for a period of less than one year. They will not be granted annual leave. Their pay will include an amount of 4/52nds of salary as vacation pay.

Service excludes all employment time with the unpaid absences in excess of four consecutive weeks with the exception of casual Union leave and Article 28.

21.2 Annual Leave Year

Employees do not require permission to carry over annual leave entitlements as long as all annual leave entitlements accrued by August 31 of any year are taken by August 31 of the next year. Employees shall advise their supervisor if they plan to carry over annual leave entitlements to the following year.

21.3 Special Circumstances

In special circumstances employees may have consideration given to annual leave arrangements outside of Article 21.2 by submitting a written application to their supervisor with a copy to the Human Resources Department.

Other than as provided for in Article 21.1, no arrangement will be made for pay in lieu of annual leave except in the case of annual leave not being taken by an employee because of a written request by the employer resulting from unusual circumstances.

21.4 Time of Annual Leave

Employees, insofar as the regular operation of the department permits, will be allowed to take their annual leave at the time they request.

The employer shall make a reasonable-effort to permit employees to take all accumulated annual leave at one time if so desired by the employee.

As far as possible, annual leave shall be arranged to secure equal opportunities of advantageous periods to all employees without regard to seniority.

21.5 Named Holiday During Annual Leave

When a named holiday falls within an employee's annual leave, such employee shall be granted one additional day of annual leave. (See Article 20)

21.6 Annual Leave for Shift Employees

Shift employees annual leave shall commence the day following an employee's weekly day or days off, unless otherwise mutually agreed.

21.7 Approved Absence During Annual Leave

Where, in respect of any period of annual leave, an employee:

- a) requires bereavement leave as per Article 30; or
- requires sick leave as per Article 29 or receives short-term disability benefits as a result of being seriously ill, seriously injured or hospitalised, or
- c) is granted other approved leave of absence;

the period of annual leave so displaced shall be either added to the employee's annual leave entitlement or reinstated for use at a later date.

21.8 Requests For Annual Leave

Requests for annual leave must be made in writing **or through the University's leave management system**. The employee is expected to provide at least two (2) weeks notice of his/her intent to take annual leave of greater than five (5) consecutive days. In all other cases the employee is expected to provide at least twenty-four (24) hours notice unless it can be shown that there is **an** extenuating circumstance.

In an emergency, an application may be made by telephone (to be confirmed in writing) where the particular circumstances warrant it. Such applications shall not be **un**reasonably withheld.

ARTICLE 22 -- EDUCATION LEAVE AND CAREER DEVELOPMENT

22.1 Recognition

The employer recognizes that educational advancement and professional development of its employees is a mutual benefit and will therefore provide career counseling, leaves and assistance as outlined below.

22.2 Career Counselling

The Director of Human Resources will, upon request, meet with an employee to discuss current and/or future career possibilities and any additional training required to prepare the employee for professional advancement. The Director of Human Resources may refer the employee to other sources for additional counselling.

22.3 Taking Classes

22.3.1 University Classes

Employees are encouraged to take classes offered by the First Nations University or the University of Regina. The work schedule of an employee may, by mutual written agreement between the employee and the supervisor, be arranged in order to permit a class to be taken during the day. Permission will not be unreasonably withheld. If permission is withheld, a written explanation will be provided by the supervisor to the employee with a copy to the Human Resources Department.

The tuition and required fees for one **credit or non-credit** class per semester at the First Nations University or the University of Regina will be paid by the employer for employees. The maximum tuition per class, paid by the employer, shall not exceed that for a three credit hour class. The maximum assistance is tuition and required fees and the cost of required textbooks.

An employee may make a written request to take a university class to the appropriate supervisor with a copy to the Human Resources Department, outlining the benefit to the employee and the employer, at least 60 days prior to the start date of the class and must be approved or denied by the supervisor within fourteen (14) days of the submitted request. The supervisor shall consider the employee's career goals and the impact the member's absence will have on the work unit in determining whether or not to approve the employee's request. Should the request not be approved, the supervisor shall provide a written explanation to the employee with a copy to the Human Resources Department.

Any employee who qualifies for educational assistance on the basis of aboriginal ancestry is required to pursue such funding prior to requesting **assistance** from the employer. In the event educational assistance on the basis of aboriginal ancestry is not approved for the employee, the employee will be required to **provide proof of registration in the class.**

The employer will pay the tuition and required fees on behalf of the employee with the understanding that if the employee does not successfully complete the course, the employer will recover tuition costs by deduction from the employee's pay.

22.3.2 Classes Other than University Classes

In place of tuition, for one class per semester at the First Nations University or the University of Regina, the employer may pay tuition, and course fees for other classes at another post-secondary institution provided the First Nations University or University of Regina does not offer an equivalent course, and it is mutually beneficial to the employee and the employer.

The employee may apply to the supervisor outlining the course, fees, dates, times and any other relevant information about the course during the semester before the class is scheduled to begin and must be approved or denied within 14 days of the submitted request. Should the request not be approved, the supervisor shall provide a written explanation to the employee with a copy to the Human Resources Department.

22.4 Education Leave

An employee having three years of continuous employment with the First Nations University may apply for education leave. The employee shall submit a written request to the supervisor, with a copy to the Union and the Human Resources Department stating the length of the leave and the nature of the educational program at least six months prior to the commencement of the leave.

Education leave may be for up to two years and the employee may apply for funding up to 80% of annual salary; however, an employee who qualifies for educational assistance on the basis of aboriginal ancestry is required to pursue

such funding first. The Human Resources Department will review the application and inform the employee at least five months prior to the commencement of the leave whether the leave has been approved and the amount of funding available. Application of this article will be fair and equitable between individuals and departments and approval will not be unreasonably withheld.

The Human Resources Department shall draw up an educational leave agreement outlining the terms and conditions of the leave including the manner of repayment if the member does not complete the terms and conditions of the leave. The agreement shall be signed by the employee, the appropriate supervisor, the Director of Human Resources and the Vice-President of Finance and Administration. Signed copies will be provided to the employee and the Union, and placed in the employee's official personnel file.

Pension and benefits provisions under Article 23 will continue during the leave with pay period, if the employee on leave continues to pay the employee's share.

Sick leave and annual leave entitlements will not accrue.

At the expiration of the leave, the employee will return to their former position and at the same pay grade as when the position was vacated and will be granted all the rights and privileges associated with employment except where otherwise stated in this Agreement.

In exceptional cases and upon agreement between the employer, the employee and the Union, the above time limits may be waived and will be confirmed in a written agreement signed by the above mentioned parties.

22.5 Professional Development Leave

The parties to this Agreement recognize that professional development is an important part of an employee's growth and career development and that the employer will support employee attendance at work-related workshops, seminars, conferences and training courses. During the annual performance review, the employee and the supervisor should review the career development of the employee and develop a systematic approach to achieving mutually agreed upon goals.

22.5.1 Employer Directed Development

Where the employer requires an employee to take a specified course, the employer will pay the cost and associated expenses of the course. When an employee requires time off to attend a course, the employer will provide time off with pay. Where required classes are taken outside an employee's normal working hours, an equivalent number of hours off work will be provided.

Employer Directed Development shall pertain to but not be limited to employee's who are required to upgrade the knowledge, skills and abilities requirements as stated on the current job description for that position.

22.5.2 Employee-Requested Development

All leaves and financial agreements must be concluded prior to the seminar, course, conference or similar program. In the event an arrangement could not be concluded prior, then the supervisor shall reasonably determine to deny or allow the leave with pay.

Employees may apply to their supervisor for permission to attend workrelated workshops, seminars, conferences and training courses, including First Nations cultural development opportunities. A written response to the request shall be made within fourteen (14) calendar days. If leave is granted, it shall be with pay.

22.6 Financial Assistance

An employee may apply for financial assistance from the employer to participate in approved development opportunities; however, an employee who qualifies for educational assistance on the basis of aboriginal ancestry is required to pursue such funding first.

Pending approval of band or Metis funding, an employee must make application for funding before June 1st of each year and the employer must render a written decision to the employee by July 15th of that same year.

The written application to the First Nations University shall include a concise budget for approval and must be submitted to the Human Resources Department along with the written recommendation and justification of the employee's supervisor.

Normally, the maximum assistance is reimbursement of registration fees and/or tuition, and assistance with book purchases and travel. Financial assistance with accommodations may or may not be considered by the employer.

ARTICLE 23 -- EMPLOYEE PENSION AND BENEFITS PLANS

The employer may only modify Employee Pension and Benefits Plans following consultation with employees through their representatives on the Joint Pension and Benefits Committee and subject to Article 2.1.

The bargaining unit shall have up to two representatives on a Joint Pension and Benefits Committee comprised of each user group covered by the Employee Pension and Benefits Plans.

The employer shall convene a Joint Pension and Benefits Committee meeting within thirty (30) days of a written request being made by a bargaining unit committee representative. A reason for a meeting must be provided in the written request and the employer will not deny a reasonable request.

All permanent employees, and term employees with appointments longer than one (1) year will be enrolled in the Employee Pension and Benefits Plans. Term employees whose appointments extend beyond one (1) year are eligible to participate in the Pension and Benefits Plans on the first pay period coinciding with or next after the date of completion of one (1) year of service. Permanent and term part-time employees are eligible to participate in the Pension and Benefits Plans on the first pay period coinciding with or next after the date of completion of one (1) year of service. (See Article 21.1 for the definition of "service".)

23.1 Pension Plan

For all employees who are eligible to join the plan, enrolment in the Group Pension Plan is a condition of employment.

23.1.1 Payment of Contributions

The employer contributes a matched amount of the employee's contribution to the Pension Plan **as follows:**

Effective January 1, 2016, pension contributions will increase for both parties to 6%.

23.1.2 Principles Governing the Pension Plan

The parties agree that the pension plan defines a form of deferred compensation that exists for the sole benefit of the members of the pension plan and their beneficiaries. Any surplus will be used for the benefit of the members and their beneficiaries.

23.2 Group Insurance

For all employees who are eligible to join the plan, participation in the Group Insurance Plan is compulsory.

23.2.1 Life Insurance/Accidental Death and Dismemberment

The employer shall offer life insurance and accidental death and dismemberment insurance at no cost to each employee.

23.2.2 Weekly Indemnity/Long-Term Disability

The employer shall offer weekly indemnity and short- and long-term disability insurance at no cost to each employee.

23.3 Dental and Extended Health and Vision Care Benefits Plans

All CUPE staff members, their spouses, and their eligible dependents shall be covered by a dental plan and an extended health and vision care benefits plan. All eligible CUPE staff members, upon hiring, shall be enrolled in the plans and continue to be enrolled unless they opt out of the plan by completing the necessary forms. The employer and the members shall share equally the cost of premiums for these plans.

The employer shall provide CUPE with information concerning premiums and claims as reported to the employer by the insurance provider. There shall be an annual meeting with representatives of the insurance provider, the employer, and CUPE to discuss issues of common concern. Members of the Pension and Benefits Committee shall be invited to this meeting.

23.4 Employee Benefits During Leave of Absence Without Pay

23.4.1 Leave Without Pay for Four Consecutive Weeks or Fewer

If the leave without pay is for four consecutive weeks or fewer, employee benefit coverage and payments by the employer remain the same.

23.4.2 Leave Without Pay for More Than Four Consecutive Weeks

In instances of weekly indemnity/long-term disability, education leave or parental leave, the employer will continue to maintain the Group Insurance Plan coverage and the employer's share of Dental and Extended Health and Vision Care Benefits premiums and Pension Plan contributions, if the employee continues to pay the employee's share.

In instances of layoff, no contributions will be made to the Pension Plan. To maintain coverage in the Group Insurance Plan and Dental and Extended Health Benefits Plans, see Article 17.7.10.

In any case, all coverage will terminate after a full twelve-month period of leave of absence without pay.

23.5 Coverage

If the following creates any compliance issues with the administration of Pension, Group Insurance, Dental or Extended Health and Vision Care Benefits Plans, the plans shall govern.

Where a benefit plan provides for a benefit to a spouse, that benefit shall be available or payable to a same sex spouse; and where the plan provides for

benefit to a child, that benefit shall be available to the child of a same sex spouse.

The meaning of spouse for benefit plan purposes shall be as defined in Section 29.3 of the Labour Standards Act. A child shall mean a child of an employee or spouse.

23.6 Annual Statement

Annually, the employer shall provide to each employee a detailed statement, which outlines in clear terms each of the benefit plans under which the employee is covered, and the benefits, which the employee derives from the plans.

23.7 Employee Family Assistance Program (EFAP)

The employer will continue to provide at no cost to all eligible employees, an Employee Family Assistance Program.

ARTICLE 24 - RE-ENTRY INTO THE WORKPLACE (AFTER LONG-TERM DISABILITY)

The employer agrees to consult with the Union on any accommodation requirements as governed by the Human Rights Code for an employee returning from disability leave.

An employee who has been away for a period greater than thirty-six (36) months shall not be entitled to return to their previous position. The employer shall give the returning employee first opportunity for the next three available positions they are sufficiently qualified to perform.

ARTICLE 25 -- INSURANCE

25.1 Liability Insurance

The employer shall maintain liability insurance to protect itself against claims arising out of actions by officers and staff members of the First Nations University of Canada acting in an authorized capacity on behalf of the University.

25.2 Travel Insurance

The employer will cover the employee with travel insurance when the employee is traveling with the authority of the employer, whether or not at University expense.

ARTICLE 26 -- LEAVE FOR COURT, INCARCERATION OR JURY DUTY

26.1 Leave for Court Appearance or Incarceration

In the event an employee is accused of an offence which requires a court appearance, the employee shall be entitled to leave of absence without pay and without loss of seniority or accrued benefits, such leave to cover time required for pre-trial legal consultation, court appearance and pre-trial legal custody.

In the event of being found guilty of an offence not involving the University, the employee may be granted leave of absence without pay to cover the period of incarceration.

26.2 Leave for Jury Duty or Court Summons

When an employee is summoned for jury duty or as a court witness, the employee shall not suffer any loss of salary or wages while so serving, except in situations where testifying on the employee's own behalf. Remuneration paid to the employee by the court must be turned over to the employer; however, this will not include expenses paid by the court.

ARTICLE 27 -- POLITICAL OFFICE, UNION AND SECONDMENT LEAVE

27.1 Leave to Seek Nomination or Election

Upon written request to their supervisor, the employer shall grant an employee leave of absence without pay to seek nomination as a candidate and to be a candidate in a municipal, provincial, federal, First Nations or Metis government.

27.1.1 Notice

The employee shall inform their supervisor when deciding to seek candidacy, shall make every attempt to give the longest possible notice and shall actively co-operate in all arrangements for the uninterrupted continuation of the employee's work.

27.1.2 Upon Election to Hold Office

Should the employee be successful in her/his bid for office, the employee may be required to resign or to take partial or complete leave without pay, depending on the requirements of the employee's responsibilities and the employer's ability to suitably replace the employee. Such resignation or leave will be effective the date he/she takes office. The leave shall be for a specific term and may be renewed. This section will be fairly and reasonably applied in light of all of the circumstances surrounding the elected position.

27.1.3 Return to Employment After Term in Office

At the expiration of the leave, the employee may return to employment with the First Nations University of Canada at the same pay level as when their position was vacated. The employee's seniority shall be retained but not accumulated during the period of absence. In the case of partial leave, seniority accumulation will be prorated.

27.1.4 Return to Employment if Election to Office Unsuccessful

Should the employee be unsuccessful in his/her bid for office, he/she shall return to the position and pay level he/she held at the time the leave was granted.

27.2 Secondment Leave

An employee selected for a temporary position with another First Nations government, agency or educational institution, may be granted leave of absence without pay for a period of up to one year. No extensions beyond one year shall be granted. The employee's seniority shall be maintained but not accumulated during such leave.

No employee will be required to accept assignment with another employer.

At the end of the secondment, the employee shall return to the position and pay level she/he vacated.

27.3 Casual Union Leave

The employer agrees that leave of absence with pay and benefits (subject to the Union providing full funding to the University) shall be given to any designated employee on leave for Union business, such leave of absence to be granted for a period not to exceed six (6) months. Requests for all leaves for Union business funded by the Union will be made by the Union in writing to the employee's supervisor with a copy to the Human Resources Department. Permission for an employee to attend to Union business will be conveyed within forty-eight (48) hours of receipt of the request, except in the case of leave for more than five (5) working days, in which case the permission will be granted within ten (10) working days of when the written request is received by the supervisor.

In addition, the employee concerned shall notify the immediate supervisor as soon as becoming aware that a request for Union leave is going to be made by the Union.

An employee on such leave shall return to their former classification, position and salary, subject to any general increases.

ARTICLE 28 - MATERNITY/ADOPTION/PARENTAL LEAVE OR CHILD IN CARE UNDER 18 MONTHS

28.1 Service Requirements for Maternity/Adoption/Parental Leave/Child in Care An employee shall qualify for maternity/adoption/parental leave (leave of absence without pay) after successful completion of the probationary period.

The employer shall not deny a pregnant employee the right to continue employment during her pregnancy provided she can supply a medical certificate as to her fitness to do so, if so requested.

28.2 Length of Maternity/Adoption/Parental Leave/Child in Care

Maternity/adoption/parental leave shall cover a period of up to fifty-two weeks in total and may be taken at the employee's discretion before and/or after the birth or adoption of a child. The leave shall normally be taken within fifty-two (52) weeks of the birth or adoption of the child.

28.3 Supplementary Employment Benefits

Provided they are in receipt of HRDC Employment Insurance Benefits, women on maternity leave, or an employee who has declared to the Human Resources Department that the employee is the primary caregiver of the child, will receive the difference between Employment Insurance Benefits received from Human Resources Development Canada and ninety-five percent of the member's salary, chargeable to the employee's accumulated sick leave benefits, while on leave for a maximum of fifteen (15) weeks, subject to the condition that the member's earnings (from employment insurance, earnings and any other source) cannot exceed one hundred percent (100%) of pre-leave earnings.

The employee's accumulated sick leave and vacation leave at the time the leave commences shall be retained to the employee's credit.

Benefits will be in accordance with Article 23.

28.4 Seniority Status During Maternity/Adoption/Parental Leave/Child in Care The employee shall continue to earn seniority during the leave.

28.5 Return from Maternity/Adoption/Parental Leave/Child in Care When an employee decides to return to work after leave, the employee shall provide the employer with at least two (2) weeks notice. On return from maternity/adoption/parental leave, the employee shall be placed in the employee's former position, classification and salary, subject to any general increases. The employee's increment dates will be set back by the total of any maternity/adoption/parental leave in excess of fifty-two (52) weeks.

ARTICLE 29 -- SICK LEAVE

29.1 Sick Leave Defined

Sick leave means the period of time an employee is absent from work with full pay by virtue of being sick or disabled, quarantined by a duly authorized medical person, or under examination or treatment by a physician, including donation of organs, chiropractor or dentist, donating blood or because of an accident for which compensation is not payable under the Workers Compensation Act.

29.2 Rate of Accumulation

Full-time employees shall earn cumulative sick leave at the rate of 1.25 days per month during each calendar year to a maximum of seventy-five (75) days in total. Regular part-time employees will earn sick leave on a prorated basis.

29.3 Application for Disability Benefits

Employees shall, as soon as practical, apply for short- and long-term disability benefits in accordance with their Group Disability Insurance Policy.

29.4 Entitlement on Termination

When an employee terminates employment, he/she shall not be entitled to a payout resulting from earned and unused sick leave.

29.5 Illness in the Family

An employee may use up to seven (7) sick leave days in a calendar year (January 1 to December 31) in the event of a crisis involving the care of a sick child, dependent grandchild, elderly parent or spouse when no alternative care is available.

29.6 Notification and Proof of Illness

In the event that an employee is unable to report to work due to illness, she/he shall make every effort to notify his/her immediate supervisor at least one half hour before their regular start time each day. A member absent for three consecutive days due to illness of either the employee or her/his child, dependent grandchild, elderly parent or spouse may be required to submit a medical certificate immediately upon return to work.

29.7 Time Off for Medical Examinations

Employees shall make every effort to schedule routine medical, dental and optical examinations during non-working hours. When an employee requires time off from work for these reasons, such time shall be charged against sick leave credits.

29.8 Absence in Excess of Sick Leave Credits

Any employee who fails to meet the requirements respecting sick leave or is absent in excess of earned sick leave credits may be considered to be absent without pay and his/her pay may be deducted to reflect such absence.

29.9 Utilization of Unearned Sick Leave Credits

Notwithstanding Article 29.2, in any calendar year, an employee may utilize a maximum of fifteen unearned sick leave days, provided that if the employee resigns or employment is terminated, she/he shall reimburse the employer by deduction from the final pay cheque for sick leave taken in excess of sick leave credits actually earned.

29.10 Utilization of Earned Sick Leave Credits While on Short Term Disability Employees shall, as soon as practical, apply for disability benefits from the

Employees shall, as soon as practical, apply for disability benefits from the Group Insurance Plan. Employees will receive their normal salary and benefits up to the amount of sick leave credits available. Sick leave usage will be prorated based on the difference between the employee's salary and the disability benefit.

29.11 Pressing Necessity Leave

'Pressing necessity' leave of up to three (3) days per year charged to sick leave will be granted to employees, at the discretion of their supervisor, to deal with urgent matters that may occur under unusual circumstances and that may include but are not limited to unforeseeable events such as fire or accidents, more dramatic personal or family situations, etc.

ARTICLE 30 -- COMPASSIONATE LEAVE

30.1 Grave Illness or Death - Immediate Family

An employee shall be permitted to take up to a total of five (5) days comprised of five (5) days with pay in instances where there is a grave illness or death of any of the following family members: spouse, common-law spouse, mother, father, sister, brother, son, daughter, grandparents, grandchildren, mother-in-law, father-in-law, sister-in-law, brother-in-law, son-in-law, daughter-in-law, or guardian.

30.2 Death of Other Family Members

An employee may be given approval by their supervisor to take up to two days with pay in instances of where there is a death of the following family members: niece, nephew, aunt or uncle.

30.3 Additional Days

If additional days are required, the days shall either be without pay or charged against **annual leave** entitlements and shall require the approval of the employee's supervisor.

30.4 Extended Leave of Absence - Care of a Family Member

Upon request, employees shall be granted a leave of absence of up to eight (8) weeks without pay to care for a family member who is seriously ill. The employee is not required to take the weeks consecutively. The employee may be eligible for Employment Insurance Compassionate Care Benefits through Service Canada.

During the leave, the employee shall continue to accumulate all benefits and seniority. Should the employee choose to make contributions for the period of the leave to the pension or benefits plan, the employer will pay the employer's contribution for the same period.

Employees applying for this leave must provide a doctor's certificate stating that a family member suffers from a serious medical condition and that the family member requires the care or support of one or more other family members.

An employee may request an extension without pay to the leave. Approval of an extension shall not be unreasonably denied. During an extended leave, the employee shall continue to accrue all benefits and seniority. Should the employee choose to make contributions for the period of the extended leave to the pension or benefits plan, the employer will pay the employer's contribution for the same period.

ARTICLE 31 -- OTHER LEAVES

31.1 Special Leave of Absence

Special leave of absence without pay shall be granted to an employee for good and sufficient reason; therefore, an employee must make a request in writing to the Human Resources Department with the recommendation of their supervisor stating the reason for such leave. The Union shall be notified of any such leaves applied for and the decision rendered.

An employee on such leave shall return to the former position and salary subject to any general increases.

Leave will not be granted to an employee to accept other gainful employment except for Secondment Leave. (See Article 27.2)

31.2 Leave for Band Elections

Employees who are, or become eligible to vote in an Indian Nation election shall be granted one (1) day's leave with pay on the day of the election.

31.3 Leave for Shift Workers

Employees with one (1) or more years' service whose shifts are subject to frequent change will be granted five (5) days unconditional leave of absence without pay per year. Such leave will not be used in conjunction with annual leave as in Article 21 or in conjunction with statutory holidays. Application must be made in writing at least three working days prior to the starting time of such leave.

31.4 Leave for Traditional Ceremonies

Upon written notice to the employee's supervisor, an employee may be granted up to seven (7) working days with pay for the purpose of participating in traditional First Nations ceremonies. Such leaves will not be unreasonably denied. The employee member shall make every effort to give the longest possible notice of such leaves.

31.5 Personal Leave

An employee is entitled to a leave of absence with pay to a maximum of two (2) working days during the year to accommodate the following:

- working at First Nations University of Canada's convocations, pow wows or conferences; or
- accepting a certificate/diploma/bachelor's degree.

31.6 Religious Holidays

The employer will attempt to accommodate the interests of employees in observation of their religious holidays.

ARTICLE 32 -- OCCUPATIONAL HEALTH AND SAFETY

32.1 Co-operation on Safety

It is agreed that the employer, the Union, the employees and all levels of supervision will co-operate fully to promote safe work practices, healthy working conditions and compliance with <u>The Saskatchewan Employment Act</u>.

The Union, through the participation of its members on the Joint Occupational Health and Safety Committee(s), will provide input into issues of occupational health and safety at the First Nations University of Canada and will assist wherever possible in the furtherance of safe conditions and practices.

32.2 Role of the Committee and Pay Provisions

The role of the Committee is to receive concerns and complaints from employees on matters affecting health and safety of the workplace, gather information pertaining to the complaints, investigate and make recommendations to management and the employees for the resolution of the complaint. Every Occupational Health and Safety Committee meeting will have official minutes recorded with copies forwarded to the Director of Human Resources, the Union and **when requested to** the appropriate Occupational Health and Safety Division of the provincial government. The Union will post the official minutes on Union bulletin boards.

The employer will consider time spent by Union members on committee work as time worked; therefore, no loss of pay shall be suffered.

32.3 Safety Measures

The *Chemical and Biological Substances* section of the Occupational Health and Safety Regulations requires the employer to "monitor the use or presence of, or a workers exposure to, any chemical substance or any biological substance that may be hazardous or harmful to the health or safety of a worker".

All employees either working with or in close proximity to any kind of hazardous product or dangerous material will be supplied with adequate and sufficient training, education, tools and safety equipment so as not to be exposed to unacceptable risks of the hazardous product or dangerous material.

Any employee may request that the employer substitute a safe or less hazardous substance for any chemical or biological substance currently in use. The Occupational Health and Safety Committee will co-operate in advising the employee(s) through the employer on the possibility of using suitable substitutions.

32.4 Educational and Training Programs

The employer will implement the required educational and training programs relating to the health and safety of employees at no cost to the employees and to be conducted during normal work time.

The employer will ensure the co-chairpersons of the Committee receive training respecting the duties and functions of the Committee. Upon receipt of reasonable written notice, committee members shall be permitted to take leave for period(s) not to exceed five (5) working days per year to attend relevant occupational health and safety training. When a committee member attends a training session conducted by Occupational Health and Safety or an approved training agency, the employee will suffer no loss of seniority or earnings.

Employees who feel they have not had opportunity for training on new equipment which they are required to operate as part of their normal duties should first discuss the issue with their supervisor and then, if necessary, with the Human Resources Department.

32.5 Safety and Health Reports, Records and Data

Reports of every accident or occurrence of an occupational disease at the work site will be provided to the appropriate Occupational Health and Safety Committee. The Committee members may request any pertinent health and safety records held by the employer, which are not confidential.

The Committee shall make quarterly reports to the employer and the Union reporting all incidents or complaints that were filed and investigated and the recommendations for resolution.

32.6 Right to Refuse

Employees have the right to refuse work that the employee feels is unsafe or will create a health hazard provided that prior to such refusal the employee has informed her/his supervisor and the Chair of the Committee of his/her opinion on the matter.

The Committee will investigate each refusal and make a decision if such refusal is warranted. The Committee shall report its findings to the Director of Human Resources, with a copy to the supervisor concerned, and if necessary, make recommendations for corrective action.

The employer may reassign the employee to other temporary duties during the Committee's investigations. Such reassignment will be with no loss in pay or benefits to the employee.

32.7 Security

The parities recognize that safety of employees, staff and students and the general public may require the installation of cameras in public access areas of the University campuses such as parking lots, walkways, building entrances, exits and hallways. Any area subject to such surveillance must be identified by a posted notice to that effect. Annually, CUPE Local 5791 will be informed of the changes in the number and/or location of security cameras and other security measures that may affect member privacy.

ARTICLE 33 -- COPYRIGHTS AND PATENTS

With respect to the issue of copyrights and patents, employees covered by this agreement will be dealt with in a manner consistent with that used for academic staff.

ARTICLE 34 -- NO STRIKES OR LOCKOUTS

34.1 No Strikes or Lockouts

The Union agrees that it will not engage in or participate in any slow down, strike, or picketing during the currency of this agreement. The University agrees that it will not lock out any members of the bargaining unit during the currency of this agreement.

34.2 Crossing Picket Lines During a Strike and Handling Goods

An employee covered by this Agreement shall have the right to refuse to cross a picket line arising out of a labour dispute or to refuse to handle goods, commodities or products of another employer who is involved in a labour dispute with that employer's employees, providing by so doing, it does not result in property damage to the University.

This refusal shall not be grounds for disciplinary action, but if the refusal results in the employee(s) not being able to perform their normal duties and other work is not available, the employee(s) may immediately be taken off payroll until once again able to perform their normal duties.

ARTICLE 35 -- DURATION AND CONTINUATION OF AGREEMENT

35.1 Duration

This Agreement shall be effective October 1, **2014** and shall remain in force until September 30, **2018**, and thereafter from year to year, unless written notice is given as provided in Article 36.3 entitled "Negotiations".

35.2 Expiration

This Agreement shall continue in force beyond the expiration date noted in Article 36.1, including during any period of negotiation, until a new Collective Agreement is ratified, except that the "no strike, no lockout" clause is not effective after the expiration date of this Agreement

35.3 Negotiations

Either party may, not less than sixty (60) days nor more than one hundred and twenty (120) days prior to the termination date of this Agreement, give notice in writing to the other party to open negotiations on a new Agreement.

ARTICLE 36 -- GENDER NEUTRAL JOB EVALUATION PLAN MAINTENANCE

36.1 Purpose

All Union positions will be classified using the Gender Neutral Job Evaluation Plan. The Gender Neutral Job Evaluation Plan is the basis for determining position placement applicable to existing positions, newly created positions or any positions that have changed.

36.2 Pay Grades

The Pay Grades established for the Gender Neutral Job Evaluation Plan are set forth in SCHEDULE A which forms part of this Collective Bargaining Agreement.

36.3 Composition of Committee

A Joint Job Evaluation Committee shall be comprised of two members named by the Union from different departments, and two members named by the employer. Each party shall also name one alternate member to the Committee who shall be allowed to attend at that party's discretion.

36.4 Time Off

The employer shall provide the Union members of the Joint Job Evaluation Committee time off with pay for the purposes of preparing for and conducting evaluations.

36.5 Resolution of Differences

If the Job Evaluation Committee cannot reach a consensus for any reason on any particular process issue or evaluation outcome, the issue will be referred to the Union/Management Committee for resolution. If the Union/Management Committee fails to resolve the issue within ten (10) working days, the issue will be referred to an arbitrator agreed upon by the Union and the employer for a final and binding decision.

36.6 New and Existing Positions

36.6.1

The University may **revise** the content of existing positions and create new positions as required by operational needs.

36.6.2

Existing classifications - title and rate of pay - shall not be eliminated or altered without prior agreement with the Union.

36.6.3

The title and rate of pay for any new classification or new position which may hereafter be established shall be subject to negotiation and agreement between the Union and the employer.

36.7 Process

36.7.1

The Employer shall submit a proposed job description for each new position or revised position to the Joint Job Evaluation Committee for

evaluation. The Joint Job Evaluation Committee shall meet within two (2) weeks to review the job description and prepare the evaluation. The results of the evaluation shall be forwarded to the Union for signature. A revised position will be re-evaluated in consultation with the employee and her/his supervisor within six (6) calendar months. A <u>new position</u> will be evaluated in consultation with the employee and her/his supervisor within nine (9) calendar months. If an employee's position is evaluated in a pay grade higher than the employee's current pay grade, their salary will be adjusted accordingly to:

- 1) The step in the payband of the higher paid classification which is next higher than the employee's current salary rate; or
- 2) The step in the higher pay grade which is a minimum of 6% higher than the employee's previous salary.

and retroactive to the first day of service in their new or revised position.

36.7.2

No employee shall suffer a reduction in current salary as a result of an evaluation outcome.

36.8 Requests for Reclassification

36.8.1 Review Process for Requests

When a permanent employee, their supervisor or the Human Resources Department, feels that a position is incorrectly classified, a written request for a classification review may be submitted to the Director of Human Resources with a copy to the Union. The Director shall respond within thirty (30) calendar days. Unless it can be clearly demonstrated to the Director of Human Resources that the responsibilities of the position have changed, a review will not be carried out until at least one (1) year has elapsed since the last review. Denial of a review by the Director of Human Resources may be grieved under the provisions of Article 6.3.

36.8.2 Conduction of the Review

The Job Evaluation Committee shall conduct the classification review and notify, in writing, the employee and his/her supervisor, and the Union of the findings. (See Article 37.7.1)

36.8.3 Salary Adjustment Upon Reclassification

Any salary adjustment will be retroactive to the first of the month in which the request for review was received by the Director of Human Resources. No employee will suffer a reduction in current salary as a result of this process.

ARTICLE 37 -- WAGE

October 1, 20141% increase to all rate schedulesOctober 1, 20151.5% increase to all rates after the above is appliedOctober 1, 20161.25% increase to all rates after the above is appliedOctober 1, 20171% increase to all rates after the above is applied

Retroactivity shall be for all CUPE members employed with the First Nations University of Canada as of the date of signing.

ARTICLE 38 -- JOB-SHARING

38.1

The position to be shared is a full-time permanent position. A permanent employee in a full-time permanent position may opt to share their position with another employee. Job sharing may include equal splitting of a position or another arrangement and may be for any length of time or an indefinite time. Both the Union and the employer must approve all arrangements in writing.

38.2

If agreement is reached between the parties that job-sharing will take place, only the vacant portion of the position to be shared will be posted.

38.3

All employee benefit plans will be available to the incumbents of job-sharing positions who meet the eligibility requirements. These will be pro-rated for their portion of employment.

38.4

If a long-term absence occurs due to illness, maternity leave, or other approved leave of absence; the other employee may cover the period of absence. If this is not acceptable to the other employee, then the vacancy may be filled on a term basis.

38.5

If the original incumbent of the position resigns or transfers, the position will be posted as a full time position. This shall not preclude the successful applicant from requesting a job-share agreement. If the other job-sharing employee resigns or transfers, the position may revert to full-time or will be posted for another job sharer as requested by the original incumbent.

38.6

In the event an employee is displaced as per Article 17.1 in a job-share, seniority will be a factor in determining the priority of placement in any

available position in the department. Employees who are not placed will be subject to the conditions of the layoff clause. In this situation, the employer is not obligated to special considerations such as an employee's interest in parttime work.

ARTICLE 39 -- JURISDICTION OF NEW AND REVISED POSITIONS

The Union and the employer agree that the following resolution process will be followed to determine Union jurisdiction where any new positions or existing positions are revised at the First Nations University of Canada.

Once the job description for the new or revised position has been created by the employer, which job description will include among other things the supervisor's title, the title of the position providing functional direction (where applicable), areas of responsibility, minimum qualifications (education, experience, technical and non-technical skills), proposed salary and hours of work, the employer will provide the job description and an organizational chart to URFA and CUPE along with a letter of intent to create or revise such position. The letter of intent shall include the employer's view as to which bargaining unit the position might be assigned, and where applicable will include the employer's view that the position is out of scope. Correspondence via e-mail will be acceptable.

URFA and CUPE will have five (5) working days from the date of receipt of the letter of intent, job description and organizational chart from the employer to either respond to the employer:

- a) agreeing with the proposed job description and Union/bargaining unit designation of the position <u>or</u>
- b) requesting further details and/or clarification from the employer of the reasons for the employer's view as to, among other things, which bargaining unit the position is out of scope and a list of questions the bargaining unit would like answered concerning the new position. At that time, either of the Unions may also request a joint meeting be held with the employer.

Where a joint meeting is requested by either Union, the meeting shall include representatives from the employer, URFA and CUPE and it will be scheduled by the employer to be held within five (5) working days of the employer receiving the request for the meeting.

If during the joint meeting the parties are unable to reach a mutual agreement regarding the scope and/or designation of the appropriate bargaining unit for the position in question, the parties agree to thereafter apply jointly to the

Saskatchewan Labour Relations Board for a decision regarding the scope of the new or revised position and the assignment of that position to the proper bargaining unit. Where both URFA and CUPE claim the position in question falls within the scope of their jurisdiction, the joint application to the Labour Relations Board shall involve all three parties, being the employer, URFA and CUPE. Should either CUPE and/or URFA not claim jurisdiction over the position in question, the non-disputing Union shall confirm its position in writing to the two remaining parties, and thereafter only the employer and the disputing Union shall proceed with the joint application to the Labour Relations Board.

The decision of the Labour Relations Board shall be final and binding on all parties to the agreement.

Both Unions will be entitled to have at least one representative in attendance with pay, at the joint meeting and the Labour Relations Board hearing.

The time limits outlined in this Agreement may be waived on a case-by-case basis by agreement of all parties.

ARTICLE 40 -- FUNCTIONAL SUPERVISION

The Union and the employer agree that the roles of functional supervision are to include and not exceed the following:

- approving attendance sheets, leave requests and time sheets;
- coordinating holiday schedules;
- prioritizing workload at the direction of the supervisor; and
- providing feedback to the supervisor pertaining to the roles of functional supervision.

Functional supervision does not include:

- performance reviews;
- direction of the workforce; or
- disciplinary measures.

All discipline, performance evaluations and work direction shall be performed by the supervisor.

Functional Supervisors are appointed by way of the Joint Job Evaluation Committee and job descriptions are to be updated to include the appointment.

	In Effect	1% Increase	1.5% Increase	1.25% Increase	1% Increase
	September,	Effective	Effective	Effective	Effective
	2014	October 1/14	October 1/15	October 1/16	October 1/17
PG 1	29,907	30206	30,659	31,042	31,352
	30,818	31,126	31,593	31,988	32,308
Up to	31,705	32,022	32,502	32,908	33,237
380	32,669	32,996	33,491	33,910	34,249
PG 2	31,183	31,495	31,967	32,367	32,691
	32,096	32,417	32,903	33,314	33,647
381-409	33,062	33,393	33,894	34,318	34,661
	34,079	34,420	34,936	35,373	35,727
PG 3	32,461	32,786	33,278	33,694	34,031
	33,453	33,788	34,295	34,724	35,071
410-438	34,470	34,815	35,337	35,779	36,137
	35,541	35,896	36,434	36,889	37,258
PG 4	33,871	34,210	34,723	35,157	35,509
	34,887	35,236	35,765	36,212	36,574
439-467	36,008	36,368	36,914	37,375	37,749
	37,338	37,711	38,277	38,755	39,143
PG 5	35,304	35,657	36,192	36,644	37,010
	36,478	36,843	37,396	37,863	38,245
468-496	37,833	38,211	38,784	39,269	39,662
	39,191	39,583	40,177	40,679	41,086
PG 6	37,024	37,394	37,955	38,429	38,813
	38,372	38,756	39,337	39,829	40,227
497-525	39,736	40,133	40,735	41,244	41,656
	41,248	41,660	42,285	42,814	43,242
PG 7	38,901	39,290	39,879	40,377	40,781
	40,283	40,686	41,296	41,812	42,230
526-554	41,849	42,267	42,901	43,437	43,871
	43,544	43,979	44,639	45,197	45,649
PG 8	40,936	41,345	41,965	42,490	42,915
	42,500	42,925	43,569	44,114	44,555
555-583	44,039	44,479	45,146	45,710	46,167
	45,786	46,244	46,938	47,525	48,000

SCHEDULE A - FIRST NATIONS UNIVERSITY OF CANADA CUPE 1975 SALARY GRID

PG 9	43,099 44,742	43,530 45,189	44,183 45,867	44,735 46,440	45,182 46,904
584-612	46,463 48,263	46,928 48,746	47,632 49,477	48,227 50,095	48,709 50,596
PG 10	45,083 46,827	45,534 47,295	46,217 48,004	46,795 48,604	47,263 49,090
613-641	48,602 50,452	49,088 50,957	49,824 51,721	50,447 52,368	50,951 52,892
Pay Grade 11	47,220 49,018 50,869	47,692 49,508 51,378	48,407 50,251 52,149	49,012 50,879 52,801	49,502 51,388 53,329
642-670	52,801	53,329	54,129	54,806	55,354
Pay Grade 12	49,436 51,336 53,270	49,930 51,849 53,803	50,679 52,627 54,610	51,312 53,285 55,293	51,825 53,818 55,846
671-699	55,278	55,831	56,668	57,376	57,950
Pay Grade 13	51,756 53,712 55,746	52,274 54,249 56,303	53,058 55,063 57,148	53,721 55,751 57,862	54,258 56,309 58,441
700-728	57,857	58,436	59,313	60,054	60,655
Pay Grade 14	54,285 56,424 58,641	54,828 56,988 59,227	55,650 57,843 60,115	56,346 58,566 60,866	56,909 59,152 61,475
729-757	60,962	61,572	62,496	63,277	63,910

 * The hourly rate may be calculated by dividing the annual salary by 52 weeks and then again by 35 hours.

(i.e.: \$40,936/52=\$787.23/week. \$787.23/35=\$22.49/hr.)

SCHEDULE B



Date Approve	ed:November 19, 2001		
Approved by:	President		
Policy Numbe	er:		
Subject:	Promoting Positive Human Relations: Policy to Prevent Harassment & Discrimination		

1. PREAMBLE:

The Saskatchewan Indian Federated College (SIFC) is committed to providing an environment that affirms and promotes First Nations cultures and values, and the dignity of human beings of diverse backgrounds and needs. The policy prohibiting harassment ensures that all members of the SIFC community—its Elders, students, staff, and faculty—have the right to participate equally in activities at the SIFC without the fear of harassment or discrimination. This right carries with it the expectation that all members of the SIFC community will conduct themselves in an appropriate and responsible manner, with due respect and regard for the rights of others. No member of the SIFC community shall cause or participate in discrimination against or harassment of another person.

2. POLICY:

With this policy statement the Saskatchewan Indian Federated College declares that it will neither tolerate nor condone any form of harassment or discrimination as defined in this document. Every reasonable and practicable effort will be made to see that no member of the SIFC community is subjected to harassment or discrimination in the SIFC environment. The SIFC, exercising due diligence, will act promptly, in a fair and unbiased manner, to investigate any reports of harassment or discrimination and take appropriate action when either is substantiated. Harassment and discrimination violate the rights of individuals and negatively affect the SIFC as a whole. Action taken to stop harassment and discrimination that has occurred will include a wide range of disciplinary measures, up to and including dismissal or expulsion.

3. SCOPE

The SIFC recognizes the fundamental importance of academic freedom, individual rights and natural justice. Nothing in this policy is to be interpreted in a way that will infringe upon academic freedom, or upon conduct between persons that is based upon mutual consent, or that will serve as a proscription of responsible human interaction. This policy is not to be interpreted, administered, or applied in such a way as to detract from the rights and obligations of those in supervisory or instructional roles to provide feedback on the performance of their employees or students in order to fulfill their responsibilities to manage, instruct, and discipline.

This policy applies to all Elders, faculty, staff, and students of the SIFC while on SIFC grounds or while performing duties for the SIFC. Contractors are required to conform to this policy as part of their contractual agreement.

Volunteers and visitors to the SIFC are expected to meet all obligations of this policy as a condition of free access to SIFC property.

An incident that involves the SIFC or SIFC-related events may be considered to be harassment or discrimination under the terms of this policy whether or not it occurs on campus and whether or not it occurs during normal working hours. Co-op programs, internships, practica, and field placements are included within the scope of this policy.

Two key pieces of legislation define the illegality of harassing or discriminatory behaviour: the 1979 Saskatchewan Human Rights Code and Regulations (as amended from time to time), and the Occupational Health and Safety Act and Regulations (1996). For further detail, readers should consult the original legislation.

All complaints must and will be taken seriously. The rights of all persons will be respected. All members of the SIFC community should use the procedures outlined in this policy to address incidents of alleged harassment or discrimination. Nevertheless, individuals have the right to refer their concerns to the SIFC Occupational Health and Safety Committee, Saskatchewan Labour–Occupational Health and Safety Division, the Human Rights Commission, the police, or may exercise any other legal rights under law. Members of bargaining units may use the appropriate grievance procedure as an alternative to the procedures described in this policy, but are encouraged to use this policy first. Nothing in this policy will limit or amend the provisions of any collective agreement.

4. HARASSMENT AND DISCRIMINATION - DEFINITIONS

Harassment and discrimination are, fundamentally, a selective denial of the basic right of each individual to dignity and respect. The SIFC defines within this section and the appendices, the types of harassment and discrimination which, taken together, constitute the definitions of all forms of harassment and discrimination included in this policy.

The standard of a reasonable person will be used for the interpretation of definitions and all other parts of this policy. It will be applied to determine what individuals ought to know, and what effect specific actions would be expected to have. The fact that words or actions offend an individual does not automatically mean that those words or actions constitute harassment. The subject and duration of the words or actions must be such that a reasonable person would find them not merely unpleasant, but severe enough to create a hostile environment or to meet another criterion of the *Definitions*.

Harassment and discrimination may be intentional or unintentional. The beliefs or intentions of an alleged harasser will not be accepted as an excuse for harassment if they do not meet the reasonable person standard, that is if a reasonable person would have known that the words or actions were likely to constitute harassment or discrimination.

<u>Harassment or Discrimination Based on a Prohibited Ground</u> is defined as any objectionable or unwanted conduct, comment, display, communication (in any form), behaviour, or practice that is based on a <u>prohibited ground</u> and:

- a. would reasonably be perceived to create a hostile, intimidating, or humiliating environment for working or learning, or constitute a threat to the health or safety of the individual; or
- has the effect of imposing burdens, obligations or disadvantages on, or has the effect of granting privileges or preference to, an individual or group of individuals.

<u>Prohibited grounds</u> are any of the following grounds specified by the Saskatchewan Human Rights Code and/or the Occupational Health and Safety Act (1996): race; creed; religion; colour; sex; sexual orientation; family status; marital status; disability; nationality; ancestry; place of origin; receipt of public assistance; physical size or weight. Age is also a prohibited ground, except where discrimination on the basis of age is permitted by law. One incident, or the aggregation of a series of incidents, may constitute harassment and/or discrimination. For a more complete definition and specific examples of what may or may not constitute harassment at the SIFC see:

- Appendix B Definition and Examples of Attitudinal Harassment
- Appendix C Definition and Examples of Sexual Harassment
- Appendix D Definitions and Examples of Racial Harassment

5. RESPONSIBILITIES OF INDIVIDUALS

- 5.1 All members of the SIFC community have the responsibility not to participate in or condone any form of harassment or discrimination and to see that any institutional authority they hold is not used in any way that could result, intentionally or unintentionally, in harassment or discrimination.
- 5.2 Any member of the SIFC community who believes that he or she has been subjected to harassment or discrimination is encouraged to handle the problem, by clearly and firmly making known to the alleged harasser that the harassment or discrimination is objectionable and must stop. It is understood that while this step is considered extremely important it may not be possible in all situations.
 - 5.3 Persons who believe that they are experiencing harassment or discrimination have the responsibility to report their situation to the SIFC, so that it can be effectively addressed.
 - 5.4 All persons who become involved in the resolution of a complaint have the responsibility to maintain confidentiality throughout this process. (See Section for a further explanation of confidentiality).

6. RESPONSIBILITIES OF SIFC ADMINISTRATORS

- 6.1 For the purposes of this policy, an administrator is defined as a President, Vice-President, Dean, Director, Registrar, SIFC Librarian, or Department Head.
- 6.2 The SIFC President has the responsibility to see that this policy remains effective.
- 6.3 SIFC administrators have the responsibility to remain alert for the warning signs of harassment and discrimination and to act promptly in contacting the SIFC Harassment Preventions Officers (HPOs). SIFC administrators shall cooperate with the HPO from the filing of the complaint to its resolution.
- 6.4 SIFC administrators have the responsibility to maintain an environment free of harassment and discrimination by acting as role models for the SIFC community. They also have the responsibility to watch closely for signs of harassment or discrimination, and to act to stop these practices when detected through the use of education and information distribution, and by reporting the situation to a HPO.

- 6.5 An SIFC administrator may initiate an action to deal with harassment and discrimination even though there has not been a written request for such action.
- 6.6 A HPO will be responsible for determining whether or not harassment and/or discrimination has occurred. Where possible and appropriate, a HPO will facilitate resolution satisfactory to a complaint of harassment or discrimination.

7. REPORTING HARASSMENT OR DISCRIMINATION

- 7.1 Any member of the SIFC community who believes that he or she has been subjected to harassment or discrimination is encouraged to handle the problem by clearly and firmly making known to the person(s) involved that their actions constitute harassment or discrimination, are objectionable, and must stop.
- 7.2 If this is unsuccessful or inappropriate, the person(s) believing that he or she is experiencing objectionable treatment should seek further guidance from the HPO. Requests for information about the policy and procedures may be made anonymously, and can be made to an HPO. The Harassment and Discrimination Prevention Policy document is posted on the SIFC Web Site.
- 7.3 In all cases where this information indicates to the person that harassment or discrimination as defined by this Policy may have occurred, the person is encouraged to seek assistance from an HPO.
- 7.4 The SIFC expect that concerns be raised in a timely manner; however, complaints may be made within one year of the alleged incident(s). Exceptions may be made in unusual circumstances.
- 7.5 Anyone who receives a complaint or allegation of harassment or discrimination is required to report this complaint to an HPO. The person to whom the report is made will refer the person(s) reporting the concerns to this policy.
- 7.6 Following consultation with an HPO, the complainant has the following options:
 - a) to proceed to the informal process;
 - b) to proceed to the formal process;
 - c) to take no further action under these procedures;
 - d) to pursue any other course of action available under law.

7.7 The complainant may withdraw the complaint at any time during the formal or informal process.

8. INFORMAL PROCESS

8.1 Initiating the Complaint

- 8.1.1 Any person who believes that s/he has been the victims of harassment or discrimination, may submit a written complaint alleging harassment or discrimination to a SIFC HPO.
- 8.1.2 The HPO shall immediately inform the designated HR Officer that a complaint has been received.
- 8.1.3 Informal complaints shall not proceed without the consent of the person or persons alleged to have been the victim of harassment or discrimination.
- 8.1.4 All informal complaints shall be documented by the HPO and forwarded to the designated HR Officer within one month.

8.2 Informal Resolution

- 8.2.1 Upon receipt of the complaint, the HPO shall attempt to resolve the complaint through any means s/he deems appropriate in the particular circumstances of the complaint, including traditional First Nations methods of conflict resolution or mediation.
- 8.2.2 The HPOs shall normally have one month within which to resolve the complaint.
- 8.2.3 If either the respondent or complainant does not wish to proceed through the informal process, or if the HPO fails to satisfactorily resolve the complaint, the HPO shall direct the parties to the formal process.

9. FORMAL PROCESS

- 9.1 The HPO shall contact the designated HR Officer in writing request that the formal process be initiated.
- 9.2 Formal complaints may be initiated by third parties, or by the HPOs, but shall not proceed through the formal process without the written consent of the person alleged to have been the victim of harassment or discrimination.

- 9.3 Within five working days of receipt of the written complaint, the HR Officer shall forward a copy to the respondent and invite a response. The respondent may, within ten working days of receipt of the complaint, submit a written response.
- 9.4 Within five working days of receipt of the written response, the HR Officer shall forward a copy to the complainant. Either party to the dispute may, within five working days of receipt of the written response which is filed with the HR Officer, submit such further information in writing as s/he deems fit. The HR Officer shall ensure that the respondent and complainant are in receipt of all information filed by the other.
- 9.5 If no written response is received from the respondent, the HR Officer may proceed with the formal hearing.
- 9.6 All written communications between the parties and the HR Officer shall be delivered personally or by registered mail.

9.7. Hearing Panel Members

- 9.7.1 Each Hearing Panel will consist of an Elder and three persons with no direct personal involvement or interest in the complaint. The Panel shall be comprised of a Chair, the designated HR Officer, and an HPO. The Panel will include at least one woman and one man.
- 9.7.2 The Chair of the Hearing Panel will be selected from a pool of persons previously appointed for this purpose by the President.

9.8 Hearings

- 9.8.1 After receiving all documentation, the Hearing Panel shall convene a formal hearing. The Chair shall decide any questions on procedure that may arise during the course of the hearing. The Hearing Panel:
 - a) may request the production of any documents, witnesses, or any other evidence it considers relevant for the purposes of the hearing;
 - b) may hear submissions, question the parties and witnesses, and examine the evidence;
 - c) shall maintain in confidence the evidence and the identity of the parties and witnesses; and
 - d) shall report its findings notwithstanding a refusal of the respondent to participate in the hearing.

- 9.8.2 The parties have the right to a fair hearing, including the right to appear and be present throughout the hearing, make representations, and have an advocate or other support person present.
- 9.8.3 The Hearing Panel will normally have one calendar month in which to complete the hearing and prepare a written report with recommendations to the President regarding the disposition of the complaint. The deliberation of the hearing panel shall be *in camera*.
- 9.8.4 The Hearing Panel's recommendations may range from complete exoneration of the respondent to disciplinary action.
- 9.8.6 A copy of the report of the Hearing Panel will be sent to both the complainant and the respondent.

9.9 Decision

- 9.9.1 Upon receipt of the complete documentation, the President shall review the file and shall consult with administrative officers as appropriate, regarding the proposed disciplinary, remedial, or preventative action.
- 9.9.2 The President shall render a decision within three weeks and inform the parties.

10. CONFIDENTIALITY AND RETALIATION

- 10.1 Any complaint of harassment or discrimination will be handled in confidence. The SIFC will not disclose the identity of the complainant or respondent, or the circumstances of the complaint, except where necessary for the purpose of investigation or corrective action, where required by law, or for the protection of individuals. All complaints will be recorded, and all subsequent investigations documented, but these records will also be held in confidence.
- 10.2 Interference with the conduct of an investigation, whether the complaint was substantiated or unsubstantiated, may result in disciplinary action by the SIFC. A breach of confidentiality by any individual with respect to a complaint may also constitute retaliation.
- 10.3 No person shall discharge, suspend, expel, intimidate, coerce, or otherwise retaliate against a person because that person, in good faith, participates in these procedures. Any such reprisal shall be considered harassment under this agreement.

- 10.4 No record of the initial complaint shall be kept in the complainant's personnel or student file unless the appropriate Officer determines that the complaint was fraudulent or malicious.
- 10.5 All information pertaining to a written complaint of harassment shall be retained in the Human Resources Office.
- 10.6 The SIFC may also take disciplinary action against any person submitting a fraudulent, malicious, or frivolous complaint of harassment of discrimination under the terms of the policy.

11. APPEALS

Any decision of an administrator concerning a respondent or a complainant who is a SIFC staff member may be appealed through the procedures set out in the appropriate collective bargaining agreement. If the complainant is a student, an appeal may be pursued in accordance with procedures to be established by the Student Discipline Committee of the Academic Council.

APPENDIX A: HARASSMENT & DISCRIMINATION PREVENTION RESOURCES

A team-based approach will be used to resolve complaints of harassment. The team consists of the Tawaw Committee, the Harassment Prevention Officers, and the designated HR official. The team members will be chosen by the SIFC to ensure that personnel with expertise in different types of harassment and discrimination are available as necessary.

Roles and Responsibilities

Harassment Preventions Officers (HPOs)

The Harassment Prevention Officers are appointed by the SIFC specifically to deal with harassment and discrimination concerns. The HPOs will receive complaints and be responsible for initial investigation, and facilitating informal resolution of complaints.

Designated Human Resources Official

The designated HR official is responsible coordination of the formal hearing process, the development and delivery of training and education, reporting, and compiling statistics.

The Tawaw Committee

The Tawaw is a group, appointed by the President with membership from all employee groups, whose purpose is to promote awareness of harassment and discrimination, monitor the effectiveness of these procedures and of the team, and to review this policy in detail at least every three years.

The Tawaw Committee is composed of the following:

An Elder,

A representative appointed by the SIFC Student Association,

A representative appointed by the SIFC Faculty Association,

A representative appointed by the SIFC Administrative, Professional, and Technical Employees,

A representative appointed by the SIFC CUPE local, and

A representative appointed by the out-of-scope employees.

APPENDIX B: DEFINITIONS AND EXAMPLES OF ATTITUDINAL HARASSMENT

Attitudinal harassment is defined as objectionable and unwanted behavior with no legitimate purpose that is not based on any prohibited ground under law but which:

- a) constitutes a clear pattern of mistreatment that would reasonably be perceived to be severe enough to create a hostile, intimidating, or humiliating environment for working or learning;
- b) constitutes the abuse of an unequal institutional power relationship or misuses authority.

The following situations, provided they are conducted in a professional manner with due regard for the dignity of all concerned, will not be defined as harassment or discrimination under this policy:

- 1. Situations that involve appropriate direction of the work force in compliance with existing Collective Agreements.
- 2. Situations that involve appropriate disciplinary action, either as permitted by Collective Agreements or through the student discipline process.
- 3. The correction, grading, assessment or appropriate criticism of any student's academic efforts, even if they are adverse or if the student does not agree.
- 4. Insistence on academic excellence or a reasonable quality of job performance.
- The free academic study, discussion and debate of controversial topics in an academic environment, including topics that may be offensive to groups or individuals.
- 6. Practices required or permitted by law or contract, such as mandatory retirement based upon age, or practices and procedures that are reasonable and *bona fide* in the circumstances.
- 7. The statement of any opinion by a person who has been legitimately asked to state their opinion.
- 8. Conduct between persons based upon full mutual consent where no institutional authority relationship exists.

APPENDIX C: DEFINITION AND EXAMPLES OF SEXUAL HARASSMENT

For the purpose of the policy, sexual harassment is defined as including, but not necessarily being limited to:

- 1. Unwanted attention of a sexually-oriented nature made by a person who knows, or ought reasonably to know, that such attention is unwanted; and/or
- 2. Implied or expressed promise of reward for complying with or submitting to sexually-oriented request or advance; and/or
- 3. Implied or expressed threat or reprisal for not complying with or submitting to a sexually-oriented request or advance; and/or
- 4. Sexually-oriented or gender-based remarks and/or behaviour directed at an individual or group, which may reasonably be perceived to create a negative psychological and emotional environment for work and/or study.

Examples of sexual harassment include, but are not limited to, behaviour such as:

- Engaging in a course of vexatious comment, such as sexist jokes, remarks, innuendoes or taunting;
- Displaying without justifiable academic reason pornographic or other sexually
 offensive or derogatory pictures or material;
- Suggestive or demeaning remarks or other verbal abuse, including unnecessary questioning about or informing others of a person's sexual orientation;
- Unnecessary and unwelcome touching;
- Leering or other sexually-oriented gestures;
- Compromising invitations, including persistent and unwelcome requests for dates;
- Demands for sexual favours;
- Sexual assault.

Sexual harassment can be either psychological or physical, or it can be both. Two types of sexual harassment may occur simultaneously; quid pro quo and poisoned environment. Quid pro quo sexual harassment refers to a threat or promise of reward being linked to an unsolicited and unwelcome sexual advance. A poisoned environment is created when physical or verbal conduct focusing on the sex or sexual orientation of an individual or group results in an intimidating, hostile or offensive environment.

People have a wide range of responses to sexually-based discrimination. Some illustrations include, but are not limited to; reduced dignity, confusion, resentment and anger, tension, low self-esteem, and a transfer of guilt from the aggressor to the victim. Further, victims may feel compelled to leave their jobs. Sexual harassment leads to a poisoned environment for work or study. The sense of isolation, persecution and distrust spreads well beyond the initial victim to encompass a broader range of people.

Although sexual harassment has typically involved a female victim, males or females can be harassed. Furthermore, although sexual harassment has often been initiated by a person in a superior position, this is not always the case; people in subordinate or equal positions may also be initiators. Thus sexual harassment may occur in a variety of situations: harassment of a student by a student; or of a staff member by a student; or of a student by a staff member.

Sexual harassment may involve persons of the opposite sex or the same sex. In determining cases of sexual harassment, it is the impact of the alleged behaviour on the complainant that is considered. In Canadian courts and arbitrations, intentions have been deemed to be irrelevant.

APPENDIX D: DEFINITION AND EXAMPLES OF RACIAL HARASSMENT

In this policy, racial harassment refers to demeaning or discriminatory treatment of individuals or groups on the basis of their race.

Racial harassment can be manifested in personal threats, attacks and insults, and in the structure of social institutions. Racial harassment may be intended or unintended. It may be the result of activity or arrangements that set out to discriminate or harm, or it may result from inadvertent action or ignorance.

Examples:

- Slurs, taunts, gestures, name-calling or innuendoes that relate to a student's or employee's racial background, whether made directly or in the presence of other students or employees;
- Discriminatory treatment of members of a racial group in performance evaluation, grading or personal interaction;
- Unwelcome banter, teasing or jokes that are racially derogatory or that depict members of a racial group in a stereotypical and demeaning manner;
- Displaying racially offensive or racially derogatory materials, picture or graffiti stating or implying a hierarchy of racial groups in status or ability that demeans a particular group(s);
- Condescension or patronizing attitudes towards members of a particular racial group;
- Reluctance or refusal to work or study with another student or employee because of their racial background;
- Attributing a student's or employee's perceived deficiencies or inabilities to their membership in a racial group;
- Unwarranted assumptions or prejudices about a student's knowledge or interests based on his or her racial background.

Examples of practices that are not racist include:

- The academic or institutional study of different cultures or races;
- The academic or institutional study of the phenomenon of racism or racist groups;
- The academic study of materials that are racist in character, such as literature originating in racist times or racist cultures.

SCHEDULE C

Parity with the U of R CUPE Employees

The parties agree that all current First Nations University CUPE 1975 Bargaining Unit Employees will receive a one-time payment calculated on 1% of payroll after the implementation of the Article 38 salary increase for October 1, 2007. Payments will be calculated by dividing the 1% in dollars by the total full-time employees. Less than full-time employees' payments will be prorated.

This one time payment is in lieu of furtherance of the Letter of Understanding -Job Evaluation Phase I & II dated October 21, 2002 and the subsequent Schedule C - Job Evaluation contained in the October 1, 2004 - September 7, 2007 Collective Agreement between the First Nations University and CUPE 1975.

SCHEDULE D

STUDENT LIBRARY WORKER POSITIONS

The parties agree:

- 1) The Student Worker Library Positions shall be within the scope of the Union effective the signing date of this Agreement.
- 2) They shall be informed they are Unionized.
- **3)** The Student Worker Library Positions have a unique way of hiring as per the Student Policy.
- **4)** The Student Workers shall accumulate seniority, which they can utilize for bidding purposes up to a maximum of three terms.
- 5) The Student Library Worker Positions shall adhere to the *Student Library Worker Program Policy*.
- 6) First Nations University Student Library Worker Program Policy The University Library hires students from mid-September to the end of April on a semester basis. The purpose for hiring these students is to complete daily tasks, for example, shelf reading, circulation, statistics, re-shelving returned materials, and other duties as required. The students work part-time evening and weekend hours. Students may also be called upon for extra hours at times when the library is short-staffed. Students who are interested in working for the library must fill out an application form for employment and attach their current class schedule and a copy of their transcripts.

Criteria:

- Must be a full-time First Nations University student (UEP students ineligible);
- 2. Must be reliable and dependable;
- 3. Must have keyboarding skills;
- 4. Must be in good physical health
- 5. Must be a self-motivator
- 6. Must have the ability to take initiative;
- 7. Must indicate an interest in library services;
- 8. Must be able to work independently;
- 9. Must be able to follow instructions;
- 10. Must be able to meet and work with the public;
- 11. Must be able to meet deadlines;
- 12. Must be goal-orientated and focused in order to meet multiple priorities in a timely, orderly fashion; proactive, energetic, and flexible in order to respond quickly and positively in a patron centered environment.

Student Library Worker Hours of Work:

Hours of work are Monday to Friday from 3:30 p.m. to 8:30 p.m.; Saturday and Sunday from 12:00 p.m. to 5:00 p.m.

Duties:

- 1) To shelve
- 2) To shelf read (ensure their shelf-reading area is correct, neat and orderly);
- To assist students with general reference, circulation of material, the use of Voyager, CD-ROMS, the microfilm reader and Xeroxing;
- To phone students who have materials on hold, inform them that the materials are available and will be held for three days. Put date and time called on the hold slip;
- 5) To track the number of students who use the library, keeping a record on the desk calendar;
- 6) To maintain vertical file system;
- 7) To maintain newspaper clipping file;
- 8) To obtain materials from other libraries for the Northern Campus Library;
- 9) Other related duties as assigned, there may be other duties assigned at the beginning of your shift.
- 10) At the end of your shift, ensure the library is clean by pushing in chairs, picking up garbage, and picking up books that may be left on the tables. Ensure that the lights, circulation terminal, CD-ROM terminal, Voyager terminals, and photocopier are turned off;
- 11) Student workers are responsible for finding a replacement if they are unable to report for their shift; if a replacement cannot be found, the student worker must report for work;
- 12) Other tasks as assigned.

Reasons for dismissal:

- 1. Repeated lateness;
- 2. Assigned tasks not completed;
- 3. Absenteeism (not showing up for work without prior notice);
- 4. Irresponsible and unreliable;
- 5. Inconsistency in work or inability to do the assigned work;
- a. A combination of the above criteria.

Procedures for Hiring Student Workers:

The procedure implemented for hiring student workers is as follows:

- Job description posted on Library doors and First Nations University Student Lounge;
- Short list and prepare for interview process;
- Interview process with library staff and Human Resources or designate;
- Check references and inform successful candidates-once interviewed, a candidate's name may be placed on a wait-list or called for back up;
- The unsuccessful candidates are also called;

• The students pay is adjusted by the amount of completed credit hours as follows:

	October 1, 2014	October 1, 2015	October 1, 2016	October 1, 2017
0 - 32 credit hours completed	13.21	13.41	13.58	13.72
33 - 64 credit hours completed	14.60	14.82	15.01	15.16
65 - 96 credit hours completed	16.01	16.25	16.45	16.61
97+ credit hours completed	18.08	18.35	18.58	18.77

• In the event a student leaves without sufficient notice of two weeks, the library does not re-hire, regardless of seniority.

SCHEDULE E

LETTERS OF UNDERSTANDING

The University and the Union agree that the following outstanding Letters and Memorandums of Understanding become part of the October 1, 2007 - September 30, 2010 Collective Agreement between the First Nations University of Canada and the Canadian Union of Public Employees, Local 1975:

- Letter of Understanding: Plant, Property and Maintenance dated June 14, 2005.
- > Letter of Understanding: CUPE 1975 Office Space at First Nations University
- > Letter of Understanding: Contracting Out
- > Letter of Understanding: Compensation of a Third Party
- > Letter of Understanding: Expedited Arbitration

LETTER OF UNDERSTANDING

BETWEEN

The FIRST NATIONS UNIVERSITY and OF CANADA - Hereinafter referred To as the "Employer"

The CANADIAN UNION OF PUBLIC EMPLOYEES LOCAL 1975 - Hereinafter Referred to as the "Union"

PLANT, PROPERTY AND MAINTENANCE JANITORS

The Union and the Employer agree "without prejudice" to the following:

- 1. Effective July 1, 2005 INAC janitorial services will be brought under FNUC administration.
- 2. Effective July 1, 2005 the janitor work for INAC offices will fall under the CUPE Collective Agreement.
- 3. Effective July 1, 2005 the current contracted employees cleaning INAC will be appointed with janitor jobs under the FNUC/CUPE Collective Agreement (three half-time equivalent positions).
- 4. Hours of work for all janitorial staff will be expanded to include shifts up to 11:00 P.M. except for the current employees/janitors Andrejan Quewezance and Lucy Snow who will remain on the 6:00 A.M. to 6:00 P.M. shift. New hires may be scheduled to work till 11:00 P.M.

IN WITNESS WHEREOF THE PARTIES HERETO HAVE CAUSED THESE PRESENCE TO BE EXECUTED THIS 14TH DAY OF JUNE, 2005.

FIRST NATIONS UNIVERSITY OF CANADA

University Representative

University Representative

CANADIAN UNION OF PUBLIC EMPLOYEES LOCAL 1975

Union Representative

Union Representative

LETTER OF UNDERSTANDING BETWEEN

and

The First Nations University of Canada – Hereinafter referred to as the "University" The Canadian Union of Public Employees Local 1975 – Hereinafter referred to as the "Union"

Provisions for Scheduled Employees Permanent Part-time/Casuals Kokhum Bea's Store

The Union and the University agree that:

- split shifts may be included in the schedules of <u>student workers</u> with the agreement to the employee in writing;
- 2) availability forms will be made available to scheduled employees, who will provide the completed form to the Store Manager upon the signing of this letter and by the last day of registration for all three semesters (Fall, Winter, Spring/Summer), amendment to availability upon fourteen (14) days notice to the Store Manager;
- 3) scheduled employees will be given fourteen (14) days notice of schedule changes;
- scheduling will reflect firstly, the seniority, followed by the availability of the employee scheduling will be based on the most available hours following the principles of seniority;
- 5) copies of schedules will be forwarded to the Union on the same day they are posted;
- 6) meal breaks and/or rest breaks will be provided to the scheduled employees as per Labour Standards and the Collective Bargaining Agreement and indicated on the posted schedules, breaks may be rescheduled as dictated by business needs;
- scheduled employees will be provided with unsocial hours of work premiums as per Article 11.8;
- 8) remaining hours available after the scheduling of permanent part-time employees will be filled by scheduled casual employees, firstly on a seniority basis followed by the availability of the employee;
- 9) notwithstanding Article 17, the Employer may utilize a reduction of hours for scheduled permanent part-time employees for the summer months (May 1 through August 31); nevertheless, a scheduled permanent part-time employee may elect layoff with recall for this time subject to the provisions of Article 16, paragraph 1. Furthermore, when a reduction of hours is necessary the reduction shall be by casuals in the first instance and scheduled permanent part-time in the second instance;
- scheduled permanent part-time employees who are laid off without a recall date, will be offered the provisions under Article 17;

In witness whereof the parties hereto have caused these presence to be executed this 8^{th} day of November A.D. 2006.

Representative

Mediator

Letter of Understanding

Between

The First Nations University of Canada – Hereinafter referred to as the "Employer" The Canadian Union of Public Employees – Hereinafter referred to as the "Union"

CUPE 1975 OFFICE SPACE AT FIRST NATIONS UNIVERSITY

The Union and the Employer agree that:

- The computer currently installed in the CUPE 1975 office housed at First Nations University, and any other future installations where the equipment is purchased through First Nations University by CUPE 1975, is the sole property of CUPE 1975 and that the employer agrees to provide reasonable technical and software support for these installations.
- The purchase of any additional software other than what is normally installed on employee computers will be purchased by CUPE 1975.
- 3) Office supplies, equipment and furniture contained in the said CUPE 1975 office are the property of First Nations University for which the use was collectively bargained under Article 3.3. This does not include the information and documents referred to in Items 4 and 5 of this Letter of Understanding.
- 4) All electronic information pertaining to Union business that is stored on said computer or transmitted by way of said computer or employee representative workstation computers, including electronic communication by way of employee accounts, is the sole property of CUPE 1975.
- All files and binders containing written documentation and ALL documents contained in the said CUPE 1975 office or temporarily stored at an employee workstation are the sole property of CUPE 1975.
- 6) In respect to Items 4 and 5 of this Letter of Understanding, the employer shall not access the said CUPE 1975 office nor use or withhold any documents or information (electronic, written, etc.) contained in the CUPE office without the consent of CUPE 1975.
- Key access to the said CUPE office will be limited to CUPE 1975 representatives (2 keys only) and a master key held by Plant, Property and Maintenance.
- The parties will enter into discussion regarding any other issues that may arise pertaining to said CUPE 1975 office.

Letter of Understanding

Between and

The First Nations University of Canada – Hereinafter referred to as the "Employer" **The Canadian Union of Public Employees** – Hereinafter referred to as the "Union"

CONTRACTING OUT

The Union and the employer agree that the employer may request an employee, who possesses the necessary valid and current journey person certificate, to perform a trade or other professional duties that would normally be contracted out. The employee will be paid an additional 20% of current salary for the time spent performing those duties. It is further agreed that any deficiencies found by licensed inspectors will be rectified by the employee. Time spent rectifying deficiencies will be at the employee's current salary.

It is understood that employees have the right to refuse requests of this nature. Prior to the performance of these duties the employer will provide a written document outlining the scope of the duties to the employee and to the Union.

LETTER OF UNDERSTANDING

BETWEEN:

and

The First Nations University of

Canada – Hereinafter referred to

as the "Employer"

The Canadian Union of Public Employees Local 1975 – Hereinafter referred to as the "Union"

COMPENSATION FROM A THIRD PARTY

When an employee is involved in an accident or any other action that involves the possibility of reimbursement for time away from work, the employee shall immediately contact the Human Resources Department to advise of the facts.

The employer may make advances to the employee pending settlement of the claim against a third party, either from sick leave or from the disability plan. Such advances will be repaid to the employer when settlement is obtained from the third party minus a prorate share of any legal fees and disbursements incurred by the employee to recover a claim provided that the employee will, at the request of the employer, agree to initiate a review through the Law Society of the reasonableness of the solicitor's account in the event the employer considers the account to be unreasonable.

Where the Human Resources Department becomes aware of the potential of a third party claim, pursuant to this clause the employee will be required to sign a written agreement to immediately repay when settlement is obtained.

Upon request of the employer, the employee will produce an affidavit setting forth the amount of compensation received from the third party.

Where the total time loss is less than ten (10) working days, the Human Resources Department may waive the right to a subrogated claim provided by this article.

LETTER OF UNDERSTANDING

BETWEEN: and

The First Nations University of

Canada - Hereinafter referred to

as the "Employer"

The Canadian Union of Public Employees Local 1975 – Hereinafter referred to as the "Union"

EXPEDITED ARBITRATION

The Parties to this Letter of Understanding agree to review this process after two (2) expedited arbitrations have occurred.

- (a) The procedures as set out herein will be used after Step 2 of the grievance procedure for all grievances, except those filed for terminations, promotions or policy issues and those otherwise mutually agreed.
- (b) Once a grievance has been referred to the expedited arbitration process, it shall be heard by an Arbitrator within sixty (60) days.
- (c) Subject to the approval set out in clause (a) hereof, the following criteria shall be used to determine appropriate grievances for expedited arbitration:
 - (i) only grievances that seek an individual settlement, i.e.: settlement applies only to the griever, would not result in a similar claim by other employees, shall have no precedential value and shall not thereafter be referred to by the parties in respect to any other matter in any other setting.
 - (ii) grievances that have limited depth regarding complex legal issues.
 - (iii) concerned with grievances that involve the interpretation and application of alleged violation of the collective agreement.
 - (iv) expedited grievance arbitration is appropriate where there is a limited range of solutions or single solution to the concern raised in the grievance.
- (d) On agreement that a case be expeditiously arbitrated, the parties will draw the arbitrator by chance from a list mutually agreed and the Arbitrator will act as a single Arbitrator on the matter.
- (e) The parties shall limit their use of representatives to the following:

Union: Staff representative or elected officer Employer: Human Resources department staff or a consultant, other than a lawyer

- (f) The representatives of the parties shall meet within five (5) working days prior to the hearing to discuss the issues including, but not limited to, the evidence, the procedure, and any other means of expediting the process.
- (g) The documents tabled with the arbitrator shall include:
 - i. collective bargaining agreement;
 - ii. grievance statement and replies;
 - iii. agreed statement of facts;
 - iv. a brief statement of each party's position and argument (one page each); and
 - v. where possible, an agreed statement as to the <u>exact</u> difference that the parties want decided.
- (h) The maximum number of cases to be scheduled in one day is two.
- (i) The <u>maximum</u> time allotted to hear each case is three (3) hours. The parties will endeavour to abide by this time limit; extensions may occur by mutual agreement.
- (j) The parties shall follow the following procedural guidelines:
 - (i) documents tabled;
 - (ii) brief opening statement by each of the parties;
 - (iii) witnesses (maximum two per party), examined, cross-examined and questioned by Arbitrator;
 - (iv) final argument (Brown and Beatty, or similar texts may be cited);
 - (v) the hearing will be conducted in an informal manner with limited objections and without concern for procedural irregularities;
 - Arbitrator <u>may</u> attempt to mediate, e.g. propose a possible resolution if the parties agree and if the case has not previously been through the mediation process;
 - (vii) Arbitrator <u>may</u> issue a verbal decision immediately. Within three (3) working days a written decision <u>shall</u> be rendered setting out the reasons which the Arbitrator deems necessary to convey a decision. Decision and reasons are limited to two pages. the decision of the single Arbitrator will be final and binding on the parties;
 - (viii) the parties will equally share the cost of fees and expenses of the Arbitrator;

- (ix) the griever and manager/supervisor who are party to the case and one representative of the Union shall be granted leave with pay to be present at the arbitration; and
- (x) the grievance may be removed from the expedited process at any time prior to the expedited hearing by mutual agreement.
- (k) The terms of this agreement may be changed at any time by mutual agreement of both parties.
- (l) The terms of requirements of this agreement may be waived by mutual agreement for any specific grievance.
- (m)Expedited arbitration awards shall not set a precedent and shall not be referred to by the parties in respect to any other matter.
- (n) All settlements of expedited cases prior to the hearing are made on a without prejudice basis and shall not be referred to by the parties in respect of any other matter.
- (o) The decision of the Arbitrator shall be final and binding and enforceable on all parties but in no event shall the Arbitrator have the power to change the collective bargaining agreement or to alter, modify, or amend its provisions. However, the Arbitrator shall have the power to dispose of any discharge or discipline grievance by any arrangements, which, in its opinion, it deems just and equitable.
- (p) Should the parties disagree as to the meaning of the decision either party may apply to the Arbitrator to clarify the decision, which it shall do within ten (10) days.
- (q) The Arbitrator may determine the financial or other arrangements to be made in the case of any suspension or demotion.

MEMORANDUM OF AGREEMENT

Between

First Nations University of Canada ("FNUniv")

And

Canadian Union of Public Employees ("CUPE") Local 5791

WHEREAS:

- A. FNUniv and CUPE (the "**Parties**") recognize that the Facilities Supervisor and Caretaker positions may fall within the definition of a supervisory employee pursuant to Section 6-1 (1) (o) of *The Saskatchewan Employment Act* (the "**Act**").
- B. Section 6-11 of the Act states, in relevant part:
 - (3) Subject to subsections (4) and (6), the board shall not include in a bargaining unit any supervisory employees.
 - (4) Subsection (3) does not apply if:
 - (a) the employer and the union make an irrevocable election to allow the supervisory employees to be in the bargaining unit; or
 - (b) the bargaining unit determined by the board is a bargaining unit comprised of supervisory employees.
 - (5) An employee who is or may become a supervisory employee:
 - (a) continues to be a member of the bargaining unit until excluded by the board or an agreement between the employer and the union; and
 - (b) is entitled to all the rights and shall fulfill all of the responsibilities of a member of the bargaining unit.
 - (6) Subsections (3) and (5) apply only on and after two years after the dates on which Subsection (3) comes into force.

The Parties agree as follows:

 The Parties see no benefit in having the Facilities Supervisor and Caretaker, who are currently members of CUPE 5791 bargaining unit under the FNUniv/CUPE Collective Agreement, moved into a separate bargaining unit.

- 2. For the purposes of section 6-11 (4) (a) of the *Act*, the Parties agree that the positions identified in the MOA shall be allowed to remain in the CUPE 5791 bargaining unit.
- 3. The Union and the employer agree to the resolution process in Article 40 of the Collective Agreement to determine union jurisdiction of any new or revised positions.

Signed on this 12 day of HISUST, 2015

On behalf of First Nations University of Canada

l

On behalf of Canadian Union of Public Employees Local 5791

SIGNING PAGE: OCTOBER 1, 2014 - SEPTEMBER 30, 2018 COLLECTIVE REEMENT BETWEEN THE FIRST NATIONS UNIVERSITY OF CANADA BARGAINING AND THE CANADIAN UNION OF PUBLIC EMPLOYEES, LOCAL 5791

IN WITNESS THEREOF the parties have caused these presents to be executed.

The First Nations University of Canada, represented by:

Chairman of the Board Date

Witness to the signature on behalf

of the University

Witness to the signature on behalf of the University

The Canadian Union of Public Employees Local 5791, represented by:

Union resident

Date

President

Date

Bargaining Committee Member

Date

CUPE National Repr esentative

CILES Date

DM/meg/vc/ts:cope491

Witness to the signature on behalf of the Union

Witness to the signature on behalf of the Union

Witness to the signature on behalf of the Union